MADISON COUNTY PURCHASING DEPARTMENT 100 North Side Square, Room 734 Huntsville, AL 35801

BID SUBMISSION CHECKLIST Bid #2017-54 Bid Opening Date: October 19, 2017

\checkmark	(Please check all completed items)	
	Signed, notarized and returned the Illegal Affidavit forms with bid.	
	Completed IFB Form, E-Verify Company ID, and signed and notarized	
	E-Verify Forms, pages 13 and 14 only	
	Filled in all blanks within the specifications	
	One original hardcopy, two (2) copies	
	Include bid bond, if requested	
	Included all requested information in bid package	
	Certifications/Licenses/Permits, if applicable	
	Certificate of Insurance(s).	
	Acknowledged the Addenda, if applicable	
	Put bid number, company name and address on the outside of the envelope(s)	
	Sealed bid response	
	Read specifications thoroughly!	

MADISON COUNTY COMMISSION 100 NORTH SIDE SQUARE HUNTSVILLE, ALABAMA 35801

INVITATION FOR BIDS

BID NUMBER: 2017-54

Madison County Commission will accept bids for Landscaping as needed, for a period of three years, not to <u>exceed \$450,000.</u> Bids will open on <u>Thursday, October 19, 2017</u> at 2:30 p.m. 7th Floor, Room 734 – Madison County Courthouse, Huntsville, Alabama (See attached specifications) Current bids are available on our website at <u>www.madisoncountyal.gov</u> <u>Each vendor submitting a bid must complete the Verified Statements Regarding Unauthorized Aliens.</u>

Madison County will accept bids for Landscaping, described by the attached specifications, "Landscaping for the Madison County Water Department," dated September 2017. Copies of this document can be obtained at <u>www.madisoncountyal.gov.</u> **The total contract amount shall not exceed \$450,000.00**. The period of performance is three years. The contract will be terminated when the total contract amount has been reached regardless of the period of performance.

The contract is to perform landscaping work after service line repairs and main break repairs. Generally, the work consists of clearing the disturbed area, grading the area back to original condition and installing either seed and straw or sod to match the surrounding vegetation. Contact Jay Craft at (256) 532-1552 for questions or addresses for typical landscaping repairs to anticipate under this contract.

Alabama Department of Agriculture and Industries Setting of Landscape Plants (SLP) permit Number _____

Alabama Board of General Contractor's License Number _

The Madison County Commission will receive sealed bids until the date and time of bid opening for item(s) meeting, exceeding, or equivalent to specifications. The Madison County Commission reserves the right to accept or reject any and all bids in whole or in part. Failure to comply with any of the requirements contained in this invitation for bids may result in the rejection of the entire bid submitted. Any attachments hereto become a part of the bid and will remain in the bid file. Brand names used in this invitation for bids are for comparison purposes only and are not to be construed as indicating any preference. Any items submitted as equivalent to or exceeding specifications must be described in detail.

All bids must be submitted on this form. No errors will be corrected after bids are opened. Bids made out in pencil will **NOT** be accepted. No Federal or State taxes are to be included or charged. It is the intent of the Madison County Commission to allow any business selling the type(s) of merchandise described, to participate in our bidding process. If any prospective vendor feels that any part of these specifications would prohibit their company from submitting a bid, or has any questions regarding this bid, you may email the Purchasing Department at <u>bidspecs@madisoncountyal.gov</u> prior to the bid opening date.

Each vendor must possess proper state, county, and city license, certification, or other requirements imposed, for engaging in the type of activity for which bids are solicited. Subsequent contracts will not be accepted from a vendor after a bid has been awarded. Vendors must include their contracts with the Invitation for Bids form prior to the bid opening.

Madison County reserves the right to require proof that the products bid are suitable for the purposes for which they are intended. Madison County also reserves the right to cancel a bid if the products received under the bid are not suitable for the purposes intended.

Vendors must sign and notarize their bid. Failure to do so may result in rejection of bid.

Madison County Commission reserves the right to waive any minor informality which is immaterial in nature, negligible, or trivial, and does not affect responsiveness.

SUBMIT BID TO: MADISON COUNTY PURCHASING

100 NORTH SIDE SQUARE ROOM 734

HUNTSVILLE, AL 35801

Vendor must show on envelope the bid number and opening date. Each numbered bid must be in a separate envelope.

All documents submitted to Madison County will be subject to Alabama's Open Records Laws (<u>Code of Alabama</u>, Title 36-12-40 and 41, as last amended). Due to the provisions of the Open Records Laws and the Competitive Bid Laws (especially <u>Code of Alabama</u>, Title 41-16-24(b)), the Madison County Commission cannot assure any Bidder that any information submitted with the bid, even though marked "Proprietary" will not be open to public inspection and copying.

I hereby affirm that I have not been in any agreement or collusion among vendors or prospective vendors in restraint of freedom of competition, by agreement to bid at a fixed price or to refrain from bidding, or otherwise.

THIS BID MUST BE NOTARIZED.	COMPANY NAME:	
Subscribed and sworn to before Me this day of	SIGNATURE: PRINT NAME:	
20	STREET ADDRESS:	
Notary Public	CITY:STATE: ZIP CODE: PHONE #: FAX #:	
Angela Anderson Madison County Purchasing Director	EMAIL ADDRESS:	

DATE: September 24, 2017

<u>ALL VENDORS PLEASE NOTE</u> <u>Addenda:</u> Bidders must periodically check the County's website for any addenda issued for this procurement. Madison County will not be responsible for a bidder's failure to acquire any addenda issued. The Bidder acknowledges receipt of the following addenda, if applicable: _

www.madisoncountyal.gov

Madison County, Alabama

Bid Package: Including Scope of Work, Instructions to Bidders, General Conditions, Special Conditions, Technical Specifications, and Sample Contract Documents

Landscaping for the Madison County Water Department

September 22, 2017

	Section	1	. Page
1.	SCOF	PE OF WORK	
	1.1	SUMMARY	. 1-1
2.	INST	RUCTIONS TO BIDDERS	
	2.1	BID DOCUMENTS	2-1
	2.2	GENERAL CONTRACTOR'S STATE LICENSING REQUIREMENTS	. 2-1
	2.3	EXAMINATION OF BID DOCUMENTS AND THE SITE OF THE WORK	. 2-1
	2.4	PRE-BID CONFERENCE	. 2-1
	2.5	EXPLANATIONS AND INTERPRETATIONS	. 2-1
	2.6	SUBSTITUTIONS	. 2-2
	2.7	PREPARATION AND DELIVERY OF BIDS	. 2-3
	2.8	WITHDRAWAL OR REVISION OF BIDS	. 2-4
	2.9	OPENING OF BIDS	. 2-5
	2.10	INCOMPLETE AND IRREGULAR BIDS	. 2-5
	2.11	BID ERRORS	. 2-5
	2.12	DISQUALIFICATION OF BIDDERS	. 2-6
	2.13	CONSIDERATION OF BIDS	. 2-6
	2.14	UNIT PRICES	. 2-6
	2.15	EXPLANATION OF BID ITEMS	. 2-7
	2.16	AWARD OF CONTRACT	. 2-8
	2.17	POWER OF ATTORNEY	2-9
3.	GENI	ERAL CONDITIONS	
	3.1	DEFINITIONS	. 3-1
	3.2	INTENT AND INTERPRETATION OF THE CONTRACT DOCUMENTS	. 3-2
	3.3	CONTRACTOR'S REPRESENTATIONS	. 3-4
	3.4	DOCUMENTS FURNISHED TO CONTRACTOR	. 3-4
	3.5	OWNERSHIP OF DRAWINGS	. 3-4
	3.6	SUPERVISION, SUPERINTENDENT AND EMPLOYEES	. 3-4
	3.7	REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS	BY
		CONTRACTOR	3-5

TABLE OF CONTENTS

3.8	SURVEYS BY CONTRACTOR	
3.9	SUBMITTALS	
3.10	DOCUMENTS AND SAMPLES AT THE SITE	3-9
3.11	"AS-BUILT" DOCUMENTS	
3.12	PROGRESS SCHEDULE	
3.13	EQUIPMENT, MATERIALS AND SUBSTITUTIONS	
3.14	SAFETY AND PROTECTION OF PERSONS AND PROPERTY	
3.15	HAZARDOUS MATERIALS	
3.16	INSPECTION OF THE WORK	
3.17	CORRECTION OF DEFECTIVE WORK	
3.18	DEDUCTIONS FOR UNCORRECTED WORK	
3.19	CHANGES IN THE WORK	
3.20	CLAIMS FOR EXTRA COST OR EXTRA WORK	
3.21	DIFFERING SITE CONDITIONS	
3.22	CLAIMS FOR DAMAGES	
3.23	DELAYS	
3.24	RESOLUTION OF CLAIMS AND DISPUTES	
3.25	OWNER'S RIGHT TO CORRECT DEFECTIVE WORK	
3.26	OWNER'S RIGHT TO STOP OR SUSPEND THE WORK	
3.27	OWNER'S RIGHT TO TERMINATE CONTRACT	
3.28	CONTACTOR'S RIGHT TO SUSPEND OR TERMINATE THE CO	ONTRACT
3.29	PROGRESS PAYMENTS	
3.30	CERTIFICATION AND APPROVALS FOR PAYMENT	
3.31	PAYMENTS WITHHELD	
3.32	SUBSTANTIAL COMPLETION	
3.33	OCCUPANCY OR USE PRIOR TO COMPLETION	
3.34	FINAL PAYMENT	
3.35	CONTRACTOR'S WARRANTY	
3.36	INDEMNIFICATION AGREEMENT	
3.37	CONTRACTOR'S AND SUBCONTRACTORS' INSURANCE	

	3.38	PERFORMANCE AND PAYMENT BONDS	
	3.39	ASSIGNMENT	
	3.40	CONSTRUCTION BY OWNER OR SEPARATE CONTRACTORS.	3-43
	3.41	SUBCONTRACTS	3-44
	3.42	PERMITS, LAWS AND REGULATIONS	3-45
	3.43	ROYALTIES, PATENTS AND COPYRIGHTS	3-45
	3.44	USE OF THE SITE	3-45
	3.45	CUTTING AND PATCHING	
	3.46	IN-PROGRESS AND FINAL CLEANUP	3-46
	3.47	LIQUIDATED DAMAGES	3-46
	3.48	USE OF FOREIGN MATERIALS	3-47
4.	SPEC	TIAL CONDITIONS	
	4.1	COORDINATION WITH WATER DEPARTMENT	
	4.2	SOURCE OF PROJECT FUNDING	
	4.3	CONTRACT ADMINISTERED BY MADISON COUNTY WATER	
		DEPARTMENT	
	4.4	OWNER'S REPRESENTATIVE	
	4.5	SPECIAL PROJECTS	
	4.6	PAYMENT AND INVOICES	
	4.7	MATERIALS SUPPLIED BY THE CONTRACTOR	
	4.8	CONTRACTOR RESPONSIBLE FOR LOCATING UTILITIES	
	4.9	DAMAGE TO PROPERTY	
	4.10	TRAFFIC CONTROL	
	4.11	NO STAGING AREA AVAILABLE	
	4.12	BUILDER'S RISK INSURANCE NOT REQUIRED	
5.	TECH	INICAL SPECIFICATIONS	5-1
	5.1	GENERAL	5-1
	5.2	SPECIFICATIONS	
	5.3	SITE CLEANUP	
APPI	ENDICE	ES	

A. FORMS

	A.1	BID FORM A-1
	A.2	BID BOND
	A.3	UNAUTHORIZED ALIENS AFFIDAVIT
	A.4	SWORN AFFIDAVIT OF EMPLOYER REGARDING UNAUTHORIZED
		ALIENS
	A.5	SWORN AFFIDAVIT OF SUBCONTRACTOR REGARDING
		UNAUTHORIZED ALIENS
B.	CONT	RACT DOCUMENTS (SAMPLE)
	B .1	CONSTRUCTION CONTRACT
	B.2	PAYMENT BOND A-10
	B.3	PERFORMANCE BOND

1. SCOPE OF WORK

1.1 SUMMARY.

(a) Scope of Services. The contractor shall perform landscaping repairs to disturbed areas after service line and main break repairs. The work will include clearing the area of rocks, roots and obstructions, grading the area back to original condition before the repair and installing either seed and straw or sod. The seed or sod should match the vegetation prior to the service line or main repair. The water main and service line repairs will be paid a fixed rate based on the type of repair.

Additionally, the contractor may be directed to perform landscaping not related to water main or service line repairs. These tasks may include installing rip rap, installing topsoil and removing and disposing of excess borrow and waste material. These tasks will be paid based on unit prices by the hour, square yard, cubic yard or ton.

The contract is an indefinite delivery and indefinite quantity contract. Madison County does not commit to any minimum amount of work being performed under the contract. The total contract value shall not exceed \$450,000. The period of performance is three years. The contract will be terminated when the total contract value has been reached regardless of the period of performance.

(b) Contents. Section 2 of this document provides the Instructions to Bidders. The General Conditions are included in Section 3. Special Conditions are provided in Section 4. The Technical Specifications for the project are described in Section 5. Appendix A includes the contract forms. Appendix B provides samples of contract documents.

2. INSTRUCTIONS TO BIDDERS

2.1 BID DOCUMENTS. The Bid Documents consist of the Advertisement for Bids, these Instructions to Bidders, any modifications of or supplements to these Instructions to Bidders, the Bid Form and the proposed Contract Documents. The Contract Documents consist of the Construction Contract, the Performance Bond and Payment Bond, the Conditions of the Contract (General, Special, and other Conditions), Technical Specifications including diagrams and drawings, and all addenda issued prior to execution of the Construction Contract. Bid Documents may be obtained or examined as set forth in the Advertisement for Bids.

2.2 GENERAL CONTRACTOR'S STATE LICENSING REQUIREMENTS. All Contractors performing work for Madison County must comply with all applicable licensing and qualification requirements including, but not limited, to Code of Alabama 34-8-1 through 34-8-28 (1975). Generally, when the amount bid for a contract exceeds \$50,000, the bidder must be licensed by the State Licensing Board for General Contractors and must provide Madison County evidence of license before bidding or the bid will not be received by or considered by Madison County, Alabama. A bid exceeding the bid limit stipulated in the bidder's license, or which is for work outside of the type or types of work stipulated in the bidder's license, will not be considered. In case of a joint venture of two or more Contractors, the amount of the bid shall be within the maximum bid limitation as set by the State Licensing Board for General Contractors of the combined limitations of the partners to the joint venture.

2.3 EXAMINATION OF BID DOCUMENTS AND THE SITE OF THE WORK. Before submitting a bid for the Work, the bidders shall carefully examine the Bid Documents, visit the site, and satisfy themselves as to the nature and location of the Work, and the general and local conditions, including weather, the general character of the site, the character and extent of existing work within or adjacent to the site, and any other work being performed thereon at the time of submission of their bids. They shall obtain full knowledge as to transportation, disposal, handling, and storage of materials, and availability of water, electric power, and all other facilities in the area which will have a bearing on the performance of the Work for which they submit their bids. The submission of a bid shall constitute a representation by the bidder that the bidder has made such examination and visit and has judged for and satisfied himself or herself as to conditions to be encountered regarding the character, difficulties, quality, and quantities of work to be performed and the material and equipment to be furnished, and as to the contract requirements involved.

2.4 PRE-BID CONFERENCE. Madison County does not plan to conduct a pre-bid conference. If required based on bidders' questions, a pre-bid conference will be scheduled via an addendum. Contact Jay Craft at (256) 532-1552 for addresses of typical landscaping repairs to anticipate under this contract.

2.5 EXPLANATIONS AND INTERPRETATIONS.

(a) Should any bidder observe any ambiguity, discrepancy, omission, or error in the drawings and specifications, or in any other bid document, or be in doubt as to the intention and meaning of these documents, the bidder should immediately report such, in writing, to Bailee Robinson, Water System Engineer, Madison County Water Department, 266 B Shields

Rd, Huntsville, Alabama 35811. Reports may also be sent via email to brobinson@madisoncountyal.gov.

(b) Clarification will be made only by written Addenda sent to all prospective bidders. Madison County will not be responsible in any manner for verbal answers or instructions regarding intent or meaning of the Bid Documents.

(c) In the case of inconsistency in the specifications, a bidder will be deemed to have included in its bid the better quality or greater quantity of the Work involved unless the bidder asked for and obtained the Owner's written clarification of the requirements before submission of a bid.

2.6 SUBSTITUTIONS.

(a) The identification of any product, material, system, item of equipment, or service in the Bid Documents by reference to a trade name, manufacturer's name, model number, etc. (hereinafter referred to as "source"), is intended to establish a required standard of performance, design, and quality and is not intended to limit competition unless the provisions of paragraph (d) below apply.

(b) When the Bid Documents identify only one or two sources, or three or more sources followed by "or approved equal" or similar wording, the bidder's proposal may be based on a source not identified but considered by the bidder to be equal to the standard of performance, design, and quality as specified; however, such substitutions must ultimately be approved by the Owner. If the bidder elects to bid on a substitution without "Pre-bid Approval" as described below, then it will be understood that proof of compliance with specified requirements is the exclusive responsibility of the bidder.

(c) When the Bid Documents identify three or more sources and the list of sources is not followed by "or approved equal" or similar wording, the bidder's proposal shall be based upon one of the identified sources, unless the bidder obtains "Pre-bid Approval" of another source as described below. Under these conditions it will be expressly understood that no product, material, system, item of equipment, or service that is not identified in the Bid Documents or granted "Pre-Bid Approval" will be incorporated into the Work unless such substitution is authorized and agreed upon through a Contract Change Order.

(d) If the Bid Documents identify only one source and expressly provide that it is an approved sole source for the product, material, system, item of equipment, or service, the bidder's proposal must be based upon the identified sole source.

(e) **Procedures for "Pre-bid Approval".** If it is desired that a product, material, system, piece of equipment, or service from a source different from those sources identified in the Bid Documents be approved as an acceptable source, application for the approval of such source must reach the hands of the Madison County at least ten days prior to the date set for the opening of bids. At Madison County's discretion, this ten day provision may be waived. The application for approval of a proposed source must be accompanied by technical data which the applicant desires to submit in support of the application. Madison County will give consideration to reports from reputable independent testing laboratories, verified experience records showing the reputation of the proposed source with previous users, evidence of reputation of the source for prompt delivery, evidence of reputation of the source for efficiency in servicing its products, or any other pertinent written information. The application to Madison County for approval of a proposed source must be accompanied by a schedule setting forth in which respects the materials or equipment submitted for consideration differ from the materials

or equipment designated in the Bid Documents. The burden of proof of the merit of the proposed substitution is upon the proposer. To be approved, a proposed source must also meet or exceed all express requirements of the Bid Documents. Approval, if granted, shall not be effective until published by Madison County in an addendum to the Bid Documents.

2.7 PREPARATION AND DELIVERY OF BIDS.

(a) Bid Form.

- (i) Bids must be submitted on the Bid Form as contained in the Bid Documents; an original hardcopy, one hardcopy and an electronic copy on a flash drive is required to be submitted.
- (ii) All information requested of the bidder on the Bid Form must be filled in. The form must be completed by typewriter or handprinted in ink.
- (iii) Where indicated by the format of the Bid Form, the bidder must specify lump sum prices in both words and figures. In case of discrepancy between the prices shown in words and in figures, the words will govern.
- (iv) All bid items requested in the Bid Form, including alternate bid prices and unit prices for separate items of the Work, must be bid. If a gross sum of bid items is requested in the Proposal Form, the gross sum shall be provided by the bidder.
- (v) The estimated quantities on the Bid Form are for use of Madison County only. Madison County makes no warranty representation or guarantee as the actual quantities
- (vi) In the space provided in the Invitation for Bids Form under "Bidder's Alabama General Contractors License", the bidder must insert his or her current general contractor's state license number, current bid limit, and type(s) of work for which bidder is licensed.
- (vii) The Bid Form shall be properly signed by the bidder. If the bidder is:
 - (1) **an individual**, that individual or his or her "authorized representative" must sign the Proposal Form;
 - (2) **a partnership,** the Proposal Form must be signed by one of the partners or an "authorized representative" of the Partnership;
 - (3) **a corporation**, the president, vice-president, secretary, or "authorized representative" of the corporation shall sign and affix the corporate seal to the Proposal Form. As used in these Instructions to Bidders, "authorized representative" is defined as a person to whom the bidder has granted written authority to conduct business in the bidder's behalf by signing and/or modifying the bid. Such written authority shall be signed by the bidder (the individual proprietor, or a member of the Partnership, or an officer of the Corporation) and shall be attached to the Proposal Form.

(viii) Interlineations, alterations or erasures on the Proposal Form must be initialed by the bidder or its "authorized representative".

(b) **Bid Guaranty.**

- (i) The Bid Form must be accompanied by a cashier's check, drawn on an Alabama bank, or a Bid Bond, executed by a surety company duly authorized and qualified to make such bonds in the State of Alabama, payable to Madison County.
- (ii) If a Bid Bond is provided in lieu of a cashier's check, the bond shall be on a standard form.
- (iii) The amount of the cashier's check or Bid Bond shall not be less than five percent (5%) of the contractor's bid, but is not required to be in an amount more than Ten Thousand Dollars (\$10,000).

(c) Other Required Documents. Each bidder must submit a copy of their General Contractor License.

(d) Unauthorized Aliens Affidavit (E-Verify Notice). Contractor's Unauthorized Aliens Affidavit shall be a part of the contact documents and shall be submitted with the bid package. Refer to Appendix A.

(e) **Delivery of Bids**.

- (i) Bids will be received until the time set, and at the location designated, in the Invitation for Bids unless notice is given of postponement. Any bid not received prior to the time set for opening bids will be rejected absent extenuating circumstances and such bids shall be rejected in all cases where received after other bids are opened.
- (ii) Each bid shall be placed, together with the bid guaranty, in a sealed envelope. On the outside of the envelope the bidder shall write in large letters "Proposal", below which the bidder shall identify the Project and the Work bid on, the name of the bidder, and the bidder's current general contractor's state license number.
- (iii) Bids may be delivered in person or by mail if ample time is allowed for delivery. When sent by mail, the sealed envelope containing the bid, marked as indicated above, shall be enclosed in another envelope for mailing. Bids, delivered in person, sent by US mail, or by delivery service, must be delivered to the Madison County Purchasing Department, 100 Northside Square, Room 734, Huntsville, Alabama 35801.

2.8 WITHDRAWAL OR REVISION OF BIDS.

(a) A bid may be withdrawn prior to the time set for opening of bids, provided a written request, executed by the bidder or the bidder's "authorized representative", is filed with the Madison County Purchasing Department prior to that time. The bid will then be returned to the bidder unopened.

(b) A bid which has been sealed in its delivery envelope may be revised by writing the change in price on the outside of the delivery envelope over the signature of the bidder or the bidder's "authorized representative". In revising the bid in this manner, the bidder must only write the amount of the change in price on the envelope and must not reveal the bid price.

(c) Written communications, signed by the bidder or its "authorized representative", to revise bids will be accepted if received by the Owner prior to the time set for opening bids. The Owner will record the instructed revision upon opening the bid. Such written communication may be by facsimile if so stipulated in Supplemental Instructions to Bidders. In revising the bid in this manner, the bidder must only write the amount of the change in price and must not reveal the bid price.

(d) Except as provided in Section 2.11 of these Instructions to Bidders, no bid shall be withdrawn, modified, or corrected after the time set for opening bids.

2.9 OPENING OF BIDS. Bids will be opened and read publicly at the time and place indicated in the Advertisement for Bids. Bidders or their authorized representatives are invited to be present.

2.10 INCOMPLETE AND IRREGULAR BIDS. A bid that is not accompanied by data required by the Bid Documents, or a bid which is in any way incomplete, may be rejected. Any bid which contains any uninitiated alterations or erasures, or any bid which contains any additions, alternate bids, or conditions not called for, or any other irregularities of any kind, will be subject to rejection. Bids may also be rejected for any of (but not limited to) the following causes

- (a) Failure to utilize bid forms provided by Madison County.
- (b) Lack of signature by an authorized representative on the bid form.
- (c) Failure to properly complete the bid.
- (d) Evidence of collusion among bidders.
- (e) Unauthorized alteration of proposal form.
- (f) Failure to submit a bid bond.
- (g) Failure to note the general contractor's license number as required.

2.11 BID ERRORS.

(a) **Errors and Discrepancies in the Proposal Form**. In case of error in the extension of prices in bids, the unit price will govern. In case of discrepancy between the prices shown in the figures and in words, the words will govern.

(b) Mistakes within the Bid. If the low bidder discovers a mistake in its bid, the low bidder may seek withdrawal of its bid without forfeiture of its bid guaranty under the following conditions:

- (i) **Timely Notice**: The low bidder must notify Madison County, Alabama in writing, within three working days after the opening of bids, that a mistake was made. This notice must be given within this time frame whether or not award has been made.
- (ii) **Substantial Mistake**: The mistake must be of such significance as to render the bid price substantially out of proportion to the other bid prices.
- (iii) Type of Mistake: The mistake must be due to calculation or clerical error, an inadvertent omission, or a typographical error which results in an erroneous sum. A mistake of law, judgment, or opinion shall not constitute a valid ground for withdrawal without forfeiture.

(iv) **Documentary Evidence**: Clear and convincing documentary evidence of the mistake must be presented to Madison County, Alabama as soon as possible, but no later than three working days after the opening of bids.

(c) Madison County's decision regarding a low bidder's request to withdraw its bid without penalty shall be made within 10 days after receipt of the bidder's evidence or by the next regular meeting of Madison County, Alabama. Upon withdrawal of bid without penalty, the low bidder shall be prohibited from (1) doing work on the project as a subcontractor or in any other capacity and (2) bidding on the same project if it is re-bid.

2.12 DISQUALIFICATION OF BIDDERS. Any bidder(s) may be disqualified from consideration for contract award for the following reasons:

(a) **Collusion**. Any agreement or collusion among bidders or prospective bidders in restraint of freedom of competition to bid at a fixed price or to refrain from bidding or otherwise shall render the bids void and shall cause the bidders or prospective bidders participating in such agreement or collusion to be disqualified from submitting further bids to Madison County, Alabama on future lettings. (See § 39-2-6, Code of Alabama 1975, for possible criminal sanctions.)

(b) Advance Disclosure. Any disclosure in advance of the terms of a bid submitted in response to an Advertisement for Bids shall render the proceedings void and require re-advertisement and rebid.

(c) Failure to Settle Other Contracts. Madison County, Alabama may reject a bid from a bidder who has not paid, or satisfactorily settled, all bills due for labor and material on other contracts in force at the time of letting.

2.13 CONSIDERATION OF BIDS.

(a) After the bids are opened and read publicly, the bid prices will be compared and the results of this comparison will be available to the public. Until the final award of the contract, however, Madison County shall have the right to reject any or all bids, and it shall have the right to waive technical errors and irregularities if, in its judgment, the bidder will not have obtained a competitive advantage and the best interests of Madison County, Alabama will be promoted.

(b) If the Bid Documents request bids for projects or parts of projects in combination or separately, the Bid Documents must include modifications of, or supplements to, these Instructions to Bidders setting forth applicable bid procedures. Award or awards will be made to the lowest responsible and responsive bidder or bidders in accordance with such bid procedures.

(c) The qualifications, ability, and responsibility of all bidders and their proposed sub-contracts will be considered in making the award. Only bids of such contractors who are licensed as required by the <u>Code of Alabama</u>, Titles 34 and 46, as amended, prior to the date of the bid opening will be considered.

2.14 UNIT PRICES.

(a) Work Bid on a Unit Price Basis. Where all, or part(s), of the planned Work is bid on a unit price basis, both the unit prices and the extensions of the unit prices constitute a basis of determining the lowest responsible and responsive bidder. In cases of error

in the extension of prices of bids, the unit price will govern. A bid may be rejected if any of the unit prices are obviously unbalanced or non-competitive.

(b) Unit Prices for Application to Change Orders. As a means of predetermining unit costs for changes in certain elements of the Work, the Bid Documents may require that the bidders furnish unit prices for those items in the Proposal Form. Unit prices for application to changes in the Work are not a basis for determining the lowest bidder. Non-competitive unit prices proposed by the successful bidder may be rejected and competitive prices negotiated by Madison County, Alabama prior to contract award. Unit prices for application to changes in the Work are not effective unless specifically included and agreed upon in the Construction Contract.

2.15 EXPLANATION OF BID ITEMS.

(a) **Summary.** The majority work performed under the contract will be landscaping repairs from water main and service line repairs. Individual work orders will be issued for each landscaping repair. The payment for landscaping repairs will be made at the Contractor's unit price per repair for each type of repair and landscape material. During the course of the contract, the Contractor may be required to perform landscaping not associated with a water main or service line repairs. This work will be handled as a special project. The Madison County Water Department Representative and the Contractor will mutually agree on the scope for each special project. The special project work will be paid at an hourly or quantity unit price rate.

(b) Landscaping Repairs: Main and Service Lines (Item 1 through 6).

- (i) Includes all labor, materials and equipment needed to perform landscaping repairs after service line repairs and main break repairs. Service line repairs will be either a short side repair or a long side repair. The landscaping repairs shall include the installation of either seed and straw or sod. The landscaping seed or sod shall match the surrounding vegetation prior to the service line or main break repair.
- (ii) Payment shall be made at the Contractor's unit price per repair for each type of repair and landscape material.
- (c) Landscaping Repairs and Special Projects: Additional Materials and Installation (Items 7 through 9
 - (i) Includes all labor, equipment and materials needed to install additional quantities of landscape material. Additional materials shall be installed at specifically assigned locations, which may or may not be repair sites. Additional materials include Bermuda sod, Bermuda seed and topsoil at approximately 4-inches thickness. This item is separate from the work required to perform landscaping repairs after service line and main breaks, Section 2.15(b).
 - (ii) Payment shall be made at the Contractor's unit price per square yard of Bermuda installed or per cubic yard of topsoil installed.

(d) Special Projects: Equipment and Operator Work (Items 10 through 12).

(i) Includes the equipment and operating costs for various types of equipment, such as a skid steer, mini-excavator and a dump truck.

The equipment and operating use is to be performed only at the designated location. Equipment and operator work is separate from repair work and will be assigned on an as-needed basis for specific landscaping work.

(ii) Payment shall be made at the Contractor's unit price per hour of equipment use required to perform the assigned work. This item is separate from the work required to perform landscaping repairs after service line and main breaks, Section 2.15(b). Payment is based on the unit price per hour on the Bid Form, Appendix A.1.

(e) Special Projects: Labor (Item 13).

- (i) Includes all labor costs for performing specific landscaping assignments, not landscaping repairs after service lines and main breaks. This item is separate from the work required to perform landscaping repairs after service line and main breaks, Section 2.15(b).
- (ii) Payment shall be made at the Contractor's unit price per hour of labor for one laborer.

(f) Special Projects: Rip Rap Installation (Item 14).

- (i) Includes all labor and equipment needed to install rip rap provided by Madison County. Rip Rap is to be installed only at the designated location.
- (ii) Payment shall be made at the Contractor's unit price per ton of rip rap installed.

(g) Special Projects: Removal of Miscellaneous Material (Item 15).

- (i) Includes all labor, equipment and materials needed to remove and dispose of unclassified borrow and/or waste material. Work is to be performed only at the designated location. The Contractor is responsible for proper disposal of the miscellaneous material.
- (ii) Payment shall be made at the Contractor's unit price per cubic yard of miscellaneous material removed.

2.16 AWARD OF CONTRACT.

(a) The contract shall be awarded to the lowest responsible and responsive bidder unless Madison County, Alabama finds that all the bids are unreasonable or that it is not in the best interest of Madison County, Alabama to accept any of the bids. A responsible bidder is one who, among other qualities determined necessary for performance, is competent, experienced, and financially able to perform the contract. A responsive bidder is one who submits a bid that complies with the terms and conditions of the Advertisement for Bids and the Bid Documents. Minor irregularities in the bid shall not defeat responsiveness.

(b) A bidder to whom award is made will be notified by email and confirmed by letter to the address shown on the Proposal Form at the earliest possible date. Unless other time frames are stipulated in Supplemental Instructions to Bidders, the maximum time frames allowed for each step of the process between the opening of bids and the issuance of an order to proceed with the Work shall be as follows

1. Award of contract by Madison 60 calendar days after the

	County Commission	opening of bids
2.	Contractor's return of the signed contract, with bonds and evidence of insurance, to Madison County, Alabama	15 calendar days after the contract has been presented to the Contractor for signature
3.	Madison County's approval of the Contractor's bonds and evidence of insurance and completion of contract execution	10 calendar days after the Contractor presents complete and acceptable documents to the Owner
4.	Notice to Proceed issued to the Contractor	Within 15 calendar days after final execution of contract by Madison County, Alabama

The time frames stated above, or as otherwise specified in the Bid Documents, may be extended by written agreement between the parties. Failure by Madison County, Alabama to comply with the time frames stated above or stipulated in Supplemental Instructions to Bidders, or agreed extensions thereof, shall be just cause for the withdrawal of the Contractor's bid and contract without forfeiture of bid security.

(c) Should the successful bidder or bidders to whom the contract is awarded fail to execute the Construction Contract and furnish acceptable Performance and Payment Bonds and satisfactory evidence of insurance within the specified period, Madison County, Alabama shall retain from the bid guaranty, if it is a cashier's check, or recover from the principal or the sureties, if the guaranty is a bid bond, the difference between the amount of the contract as awarded and the amount of the bid of the next lowest responsible and responsive bidder. If no other bids are received, the full amount of the bid guaranty shall be so retained or recovered as liquidated damages for such default. Any sums so retained or recovered shall be the property of Madison County, Alabama.

(d) All bid guaranties, except those of the three lowest bona fide bidders, will be returned immediately after bids have been checked, tabulated, and the relation of the bids established. The bid guaranties of the three lowest bidders will be returned as soon as the contract bonds and the contract of the successful bidder have been properly executed and approved. When the award is deferred for a period of time longer than 45 days after the opening of the bids, all bid guaranties, except those of the potentially successful bidders, shall be returned. If no award is made within the specified period, as it may by agreement be extended, all bids will be rejected, and all guaranties returned. If any potentially successful bidder agrees in writing to a stipulated extension in time for consideration of its bid and its bid was guaranteed with a cashier's check, Madison County, Alabama may permit the potentially successful bidder to substitute a satisfactory bid bond for the cashier's check.

2.17 POWER OF ATTORNEY. Attorneys-in-fact who sign bid bonds or contract bonds must file with each bond a certified and effectively dated copy of their power of attorney.

3. GENERAL CONDITIONS

3.1 DEFINITIONS. Whenever the following terms, or pronouns in place of them, are used in the Contract Documents, the intent and meaning shall be interpreted as follows:

(a) **Contract**: The Contract is the embodiment of the Contract Documents. The Contract represents the entire and integrated agreement between the Owner and Contractor and supersedes any prior written or oral negotiations, representations, or agreements that are not incorporated into the Contract Documents. The Contract may be amended only by a Contract Change Order or a Modification to the Construction Contract. The contractual relationship which the Contract creates between the Owner and the Contractor extends to no other persons or entities. The Contract consists of the following Contract Documents, including all additions, deletions, and modifications incorporated therein before the execution of the Construction Contract:

- (i) Contract
- (ii) Performance and Payment Bonds
- (iii) Conditions of the Contract (General, Supplemental, Special and other Conditions)
- (iv) Specifications
- (v) Drawings
- (vi) Contract Change Orders
- (vii) Modifications to the Construction Contract

(b) Contract Sum: The Contract Sum is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents. The term "Contract Sum" means the Contract Sum stated in the Construction Contract as may have been increased or decreased by Change Order(s) in accordance with the Contract Documents.

(c) Contract Time: The Contract Time is the period of time in which the Contractor must complete the Work. The date on which the Contract Time begins is specified in the written Notice to Proceed issued to the Contractor by the Owner. The term "Contract Time" means the Contract Time stated in the Construction Contract as may have been extended by Change Order(s) in accordance with the Contract Documents. The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

(d) Contractor: The Contractor is the person or persons, firm, partnership, joint venture, association, corporation, cooperative, limited liability company, or other legal entity, identified as such in the Construction Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

- (e) **Defective Work**: The term "Defective Work" shall apply to:
 - (i) Any product, material, system, equipment, or service, or its installation or performance, which does not conform to the requirements of the Contract Documents,
 - (ii) in-progress or completed Work the workmanship of which does not conform to the quality specified or, if not specified, to the quality produced by skilled workers performing work of a similar nature on similar projects in the state,
 - (iii) Substitutions and deviations not properly submitted and approved or otherwise authorized,

- (iv) Temporary supports, structures, or construction which will not produce the results required by the Contract Documents, and
- (v) Materials or equipment rendered unsuitable for incorporation into the Work due to improper storage or protection.

(f) **Drawings**: The Drawings are the portions of the Contract Documents showing graphically the design, location, layout, and dimensions of the Work, in the form of plans, elevations, sections, details, schedules, and diagrams.

(g) Notice to Proceed: A Notice to Proceed is issued by the Owner, fixing the date on which the Contractor shall begin the prosecution of the Work, which is also the date on which the Contract Time shall begin.

(h) **Owner**: The Owner is the entity identified as such in the Construction Contract and is referred to throughout the Contract Documents as if singular in number. The term "Owner" means the Owner or the Owner's authorized representative. The term "Owner" as used herein shall be synonymous with the term "Awarding Authority" as defined and used in Title 39 - Public Works, Code of Alabama, 1975, as amended.

(i) **The Project**: The Project is the total construction of which the Work required by these Contract Documents may be the entirety or only a part with other portions to be constructed by the Owner or separate contractors.

(j) **Specifications**: The Specifications are that portion of the Contract Documents which set forth in writing the standards of quality and performance of products, equipment, materials, systems, and services and workmanship required for acceptable performance of the Work.

(k) **Subcontractor**: A Subcontractor is a person or entity who is undertaking the performance of any part of the Work by virtue of a contract with the Contractor. The term "Subcontractor" means a Subcontractor or its authorized representatives.

(1) **The Work**: The Work is the construction and services required by the Contract Documents, and includes all labor, materials, supplies, equipment, and other items and services as are necessary to produce the required construction and to fulfill the Contractor's obligations under the Contract. The Work may constitute the entire Project or only a portion of it.

3.2 INTENT AND INTERPRETATION OF THE CONTRACT DOCUMENTS.

(a) Intent. It is the intent of the Contract Documents that the Contractor shall properly execute and complete the Work described by the Contract Documents, and unless otherwise provided in the Contract, the Contractor shall provide all labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work, in full accordance with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

(b) Complementary Documents. The Contract Documents are complementary. If Work is required by one Contract Document, the Contractor shall perform the Work as if it were required by all of the Contract Documents. However, the Contractor shall be required to perform Work only to the extent that is consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

(c) Order of Precedence. Should any discrepancy arise between the various elements of the Contract Documents, precedence shall be given to them in the following order

unless to do so would contravene the apparent Intent of the Contract Documents stated in preceding Paragraph (a):

- The Construction Contract. (i)
- (ii) Addenda, with those of later date having precedence over those of earlier date.
- General Conditions of the Contract. (iii)
- Supplementary or Special Conditions (or other Conditions which (iv) modify the General Conditions of the Contract).
- The Specifications. (v)
- Details appearing on the Drawings; large scale details shall take (vi) precedence over smaller scale details.
- The Drawings; large scale drawings shall take precedence over (vii) smaller scale drawings.

Organization. Except as may be specifically stated within the technical (**d**) specifications, neither the organization of the Specifications into divisions, sections, or

otherwise, nor any arrangement of the Drawings shall control how the Contractor subcontracts portions of the Work or assigns Work to any trade.

Interpretation. **(e)**

- The Contract Documents shall be governed by and construed in (i) accordance with the laws of the State of Alabama.
- (ii) The Contract Documents shall be interpreted collectively, each part complementing the others and consistent with the Intent of the Contract Documents stated in preceding Paragraph (a). Unless an item shown or described in the Contract Documents is specifically identified to be furnished or installed by the Owner or others or is identified as "Not In Contract" ("N.I.C."), the Contractor's obligation relative to that item shall be interpreted to include furnishing, assembling, installing, finishing, and/or connecting the item at the Contractor's expense to produce a product or system that is complete, appropriately tested, and in operative condition ready for use or subsequent construction or operation of the Owner or separate contractors. The omission of words or phases for brevity of the Contract Documents, the inadvertent omission of words or phrases, or obvious typographical or written errors shall not defeat such interpretation as long as it is reasonably inferable from the Contract Documents as a whole.
- Words or phrases used in the Contract Documents which have (iii) well-known technical or construction industry meanings are to be interpreted consistent with such recognized meanings unless otherwise indicated.
- Except as noted otherwise, references to standard specifications or (iv) publications of associations, bureaus, or organizations shall mean the latest edition of the referenced standard specification or publication as of the date of the Invitation for Bids.
- (v) In the case of inconsistency between Drawings and Specifications or within either document not clarified by addendum, the better

quality or greater quantity of Work shall be provided in accordance with the Owner's interpretation.

- (vi) Generally, portions of the Contract Documents written in longhand take precedence over typed portions, and typed portions take precedence over printed portions.
- (vii) Any doubt as to the meaning of the Contract Documents or any obscurity as to the wording of them shall be promptly submitted in writing to the Owner for written interpretation, explanation, or clarification.
- (viii) Severability. The partial or complete invalidity of any one or more provision of this Contract shall not affect the validity or continuing force and effect of any other provision.

3.3 CONTRACTOR'S REPRESENTATIONS. By executing the Contract the Contractor represents to the Owner:

(a) The Contractor has visited the site of the Work to become familiar with local conditions under which the Work is to be performed and to evaluate reasonably observable conditions as compared with requirements of the Contract Documents.

(b) The Contractor shall use its best skill and attention to perform the Work in an expeditious manner consistent with the Contract Documents.

(c) The Contractor is an independent contractor and in performance of the Contract remains and shall act as an independent contractor having no authority to represent or obligate the Owner in any manner unless authorized by the Owner in writing.

3.4 DOCUMENTS FURNISHED TO CONTRACTOR. Unless otherwise provided in the Contract Documents, five (5) sets of Drawings and Technical Specifications will be furnished to the Contractor by the Owner without charge. Other copies requested will be furnished at reproduction cost.

3.5 OWNERSHIP OF DRAWINGS. All original or duplicated Drawings, Specifications, and other documents prepared by the Owner, and furnished to the Contractor are the property of the Owner and are to be used solely for this Project and not to be used in any manner for other work. Upon completion of the Work, all copies of Drawings and Specifications, with the exception of the Contractor's record set, shall be returned or accounted for by the Contractor to the Owner, on request.

3.6 SUPERVISION, SUPERINTENDENT AND EMPLOYEES.

(a) Supervision and Construction Methods.

(i) The term "Construction Methods" shall be interpreted as the construction means, methods, techniques, sequences, and procedures utilized by the Contractor in performing the Work. The Contractor is solely responsible for supervising and coordinating the performance of the Work, including the selection of Construction Methods, unless the Contract Documents give other specific instructions concerning these matters.

- (ii) The Contractor is solely and completely responsible for job site safety, including the protection of persons and property in accordance with Section 3.14.
- (iii) The Contractor shall be responsible to the Owner for acts and omissions of not only the Contractor and its agents and employees, but all persons and entities, and their agents and employees, who are performing portions of the Work for or on behalf of the Contractor or any of its Subcontractors.
- (iv) The Contractor shall be responsible to inspect the in-progress and completed Work to verify its compliance with the Contract Documents and to insure that any element or portion of the Work upon which subsequent Work is to be applied or performed is in proper condition to receive the subsequent Work.

(b) Superintendent.

- (i) The Contractor shall employ and maintain a competent level of supervision for the performance of the Work at the Project site, including a superintendent who shall:
 - (1) Have full authority to receive instructions from the Owner and to act on those instructions; and
 - (2) Be present at the Project site at all times during which Work is being performed.
- (ii) Before beginning performance of the Work, the Contractor shall notify the Owner of the name and qualifications of its proposed superintendent so that the Owner may review the individual's qualifications. If, for reasonable cause, the Owner refuses to approve the individual, or withdraws its approval after once giving it, the Contractor shall name a different superintendent for the Owner's review and approval. Any disapproved superintendent will not perform in that capacity thereafter at the Project site.

(c) **Employees**. The Contractor shall permit only fit and skilled persons to perform the Work. The Contractor shall enforce safety procedures, strict discipline, and good order among persons performing the Work. The Contractor will remove from its employment on the Project any person who deliberately or persistently produces non-conforming Work or who fails or refuses to conform to reasonable rules of personal conduct contained in the Contract Documents or implemented by the Owner and delivered to the Contractor in writing during the course of the Work.

3.7 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR.

(a) In order to facilitate assembly and installation of the Work in accordance with the Contract Documents, before starting each portion of the Work, the Contractor shall examine and compare the relevant Contract Documents, and compare them to relevant field measurements made by the Contractor and any conditions at the site affecting that portion of the Work.

(b) If the Contractor discovers any errors, omissions, or inconsistencies in the Contract Documents, the Contractor shall promptly report them to the Owner as a written request

for information that includes a detailed statement identifying the specific Drawings or Specifications that are in need of clarification and the error, omission, or inconsistency discovered in them.

- (i) The Contractor shall not be expected to act as a licensed design professional and ascertain whether the Contract Documents comply with applicable laws, statutes, ordinances, building codes, and rules and regulations, but the Contractor shall be obligated to promptly notify the Owner of any such noncompliance discovered by or made known to the Contractor. If the Contractor performs Work without fulfilling this notification obligation, the Contractor shall pay the resulting costs and damages that would have been avoided by such notification.
- (ii) The Contractor shall not be liable to the Owner for errors, omissions, or inconsistencies that may exist in the Contract Documents, or between the Contract Documents and conditions at the site, unless the Contractor knowingly fails to report a discovered error, omission, or inconsistency to the Owner, in which case the Contractor shall pay the resulting costs and damages that would have been avoided by such notification.

(c) If the Contractor considers the Owner's response to a request for information to constitute a change to the Contract Documents involving additional costs and/or time, the Contractor shall follow the procedures of Section 3.20, Claims for Extra Cost or Extra Work.

(d) If, with undue frequency, the Contractor requests information that is obtainable through reasonable examination and comparison of the Contract Documents, site conditions, and previous correspondence, interpretations, or clarifications, the Contractor shall be liable to the Owner for reasonable charges from the Owner for the additional services required to review, research, and respond to such requests for information.

3.8 SURVEYS BY CONTRACTOR.

(a) The Contractor shall provide competent engineering services to assure accurate execution of the Work in accordance with the Contract Documents. The Contractor shall verify the figures given for the contours, approaches, and locations shown on the Drawings before starting any Work and be responsible for the accuracy of the finished Work. Without extra cost to the Owner, the Contractor shall engage a licensed surveyor if necessary to verify boundary lines, keep within property lines, and shall be responsible for encroachments on rights or property of public or surrounding property owners.

(b) The Contractor shall establish all base lines for the location of the principal components of the Work and make all detail surveys necessary for construction, including grade stakes, batter boards and other working points, lines, and elevations. If the Work involves alteration of or addition to existing structures or improvements, the Contractor shall locate and measure elements of the existing conditions as is necessary to facilitate accurate fabrication, assembly, and installation of new Work in the relationship, alignment, and/or connection to the existing structure or improvement as is shown in the Contract Documents.

3.9 SUBMITTALS.

(a) Where required by the Contract Documents, the Contractor shall submit shop drawings, product data, samples, and other information (hereinafter referred to as Submittals) to the Owner for the purpose of demonstrating the way by which the Contractor proposes to conform to the requirements of the Contract Documents. Submittals which are not required by the Contract Documents may be returned by the Owner without action.

(b) The Contractor shall be responsible to the Owner for the accuracy of its Submittals and the conformity of its submitted information to the requirements of the Contract Documents. Each Submittal shall bear the Contractor's approval, evidencing that the Contractor has reviewed and found the information to be in compliance with the requirements of the Contract Documents. Submittals which are not marked as reviewed and approved by the Contractor may be returned by the Owner without action.

(c) The Contractor shall prepare and deliver its submittals to the Owner sufficiently in advance of construction requirements and in a sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors. In coordinating the Submittal process with its construction schedule, the Contractor shall allow sufficient time to permit adequate review by the Owner.

(d) By approving a Submittal the Contractor represents not only that the element of Work presented in the Submittal complies with the requirements of the Contract Documents, but also that the Contractor has:

- (i) Found the layout and/or dimensions in the Submittal to be comparable with those in the Contract Documents and other relevant Submittals and has made field measurements as necessary to verify their accuracy, and
- (ii) Determined that products, materials, systems, equipment, and/or procedures presented in the Submittal are compatible with those presented, or being presented, in other relevant Submittals and with the Contractor's intended Construction Methods.

(e) The Contractor shall not fabricate or perform any portion of the Work for which the Contract Documents require Submittals until the respective Submittals have been approved by the Owner.

(f) In the case of a resubmission, the Contractor shall direct specific attention to all revisions in a Submittal. The Owner's approval of a resubmission shall not apply to any revisions that were not brought to the Owner's attention.

(g) If the Contract Documents specify that a Submittal is to be prepared and sealed by a registered architect or licensed engineer retained by the Contractor, all drawings, calculations, specifications, and certifications of the Submittal shall bear the Alabama seal of registration and signature of the registered/licensed design professional who prepared them or under whose supervision they were prepared. The Owner shall be entitled to rely upon the adequacy, accuracy and completeness of such a Submittal, provided that all performance and design criteria that such Submittal must satisfy are sufficiently specified in the Contract Documents. The Owner will review, approve, or take other appropriate action on such a Submittal only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance or design criteria specified in the Contract Documents.

(h) **Deviations**.

- (i) Madison County inspectors are authorized by the Owner to approve "minor" deviations from the requirements of the Contract Documents. "Minor" deviations are defined as those which are in the interest of the Owner, do not materially alter the quality or performance of the finished Work, and do not affect the cost or time of performance of the Work. Deviations which are not "minor" may be authorized only by the Owner through the Change Order procedures of Section 3.19.
- (ii) Any deviation from the requirements of the Contract Documents contained in a Submittal shall be clearly identified as a "Deviation from Contract Requirements" (or by similar language) within the Submittal and, in a letter transmitting the Submittal to the Owner, the Contractor shall direct the Owner's attention to, and request specific approval of, the deviation. Otherwise, the Owner's approval of a Submittal does not constitute approval of deviations from the requirements of the Contract Documents contained in the Submittal.
- (iii) The Contractor shall bear all costs and expenses of any changes to the Work, changes to work performed by the Owner or separate contractors required to accommodate an approved deviation unless the Contractor has specifically informed the Owner in writing of the required changes and a Change Order has been issued authorizing the deviation and accounting for such resulting changes and costs.

(i) Owner's Review and Approval.

- (i) The Owner will review the Contractor's Submittals for conformance with requirements of, and the design concept expressed in, the Contract Documents and will approve or take other appropriate action upon them. This review is not intended to verify the accuracy and completeness of details such as dimensions and quantities, or to substantiate installation instructions or performance of equipment or systems, all of which remain the responsibility of the Contractor. However, the Owner shall advise the Contractor of any errors or omissions which the Owner may detect during this review. The Owner's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- (ii) The Owner will review and respond to all Submittals with reasonable promptness to avoid delay in the Work or in the activities of the Owner, Contractor, or separate contractors, while allowing sufficient time to permit adequate review.
- (iii) No corrections or changes to Submittals indicated by the Owner will be considered as authorizations to perform Extra Work. If the Contractor considers such correction or change of a Submittal to require Work which differs from the requirements of the Contract

Documents, the Contractor shall promptly notify the Owner in writing in accordance with Section 3.20, Claims for Extra Cost or Extra Work.

(j) Conformance with Submittals. The Work shall be constructed in accordance with approved Submittals.

3.10 DOCUMENTS AND SAMPLES AT THE SITE.

(a) "As Issued" Set. The Contractor shall maintain at the Project site, in good order, at least one copy of all Addenda, Change Orders, supplemental drawings, written directives and clarifications, and approved Submittals intact as issued, and an updated construction schedule.

(b) "Posted" Set. The Contractor shall maintain at the Project site, in good order, at least one set of the Drawings and Project Manual into which the Contractor has "posted"(incorporated) all Addenda, Change Orders, supplemental drawings, clarifications, and other information pertinent to the proper performance of the Work. The Contractor shall assure that all sets of the Drawings and Project Manuals being used by the Contractor, Subcontractors, and suppliers are "posted" with the current information to insure that updated Contract Documents are used for performance of the Work.

(c) The documents and samples required by this Section to be maintained at the Project site shall be readily available to the Owner and their representatives.

3.11 "AS-BUILT" DOCUMENTS. Unless otherwise provided in the Contract Documents, the Contractor shall deliver two (2) sets of "As-built" documents, as described herein, to the Owner upon completion of the Work. Each set of "As-built' documents shall consist of a copy of the Record of Coatings Applied, Record of Repairs, and Project Manual, in like-new condition, into which the Contractor has neatly incorporated all Addenda, Change Orders, supplemental drawings, clarifications, field changes, corrections, selections, and other information as required herein or specified elsewhere in the Contract Documents.

(a) **Drawings**

- (i) To the greatest extent practicable, information shall be carefully drawn and lettered, in ink, on the Drawings in the form of sketches, details, plans, notes, and dimensions as required to provide a fully dimensioned record of Work. When required fort clarity, sketches, details, or partial plans shall be on supplemental sheets and bound into the Drawings and references on the drawing being revised.
- (ii) Any entry of information made in the Drawings that is the result of an Addendum or Change Order, shall identify the Addendum or Change Order from which it originated.

(b) **Project Manual**

- (i) A copy of all Addenda and Change Orders, excluding drawings thereof, shall be bound in the front of the Project Manual.
- (ii) Where a document, form, or entire specification section is revised, the latest issue shall be bound into the Project Manual in the place of the superseded issue.

(iii) Where information within a specification section is revised, the deleted or revised information shall be drawn through in ink and an adjacent note added identifying the Addendum or Change Order containing the revised information.

(c) Within ten days after the Date of Substantial Completion of the Work, or the last completed portion of the Work, the Contractor shall submit the "As-built" documents to the Owner for approval. If the Owner requires that any corrections be made, the documents will be returned in a reasonable time for correction and resubmission.

3.12 PROGRESS SCHEDULE.

(a) The Contractor, shall within fifteen days after the date of commencement stated in the Notice to Proceed, or such other time as may be provided in the Contract Documents, prepare and submit to the Owner for review and approval a practicable construction schedule informing the Owner of the order in which the Contractor plans to carry on the Work within the Contract Time. The Owner's review and approval of the Contractor's construction schedule shall be only for compliance with the specified format, Contract Time, and suitability for monitoring progress of the Work and shall not be construed as a representation that the Owner has analyzed the schedule to form opinions of sequences or durations of time represented in the schedule.

(b) If a schedule format is not specified elsewhere in the Contract Documents, the construction schedule shall be prepared using Alabama Building Commission (ABC) Form C-11, "Progress Schedule and Report", or similar format of suitable scale and detail to indicate the percentage of Work scheduled to be completed at the end of each month. At the end of each month the Contractor shall enter the actual percentage of completion on the construction schedule submit two copies to the Owner, and attach one copy to each copy of the monthly Application for Payment. The construction schedule shall be revised to reflect any agreed extensions of the Contract Time or as required by conditions of the Work.

(c) If a more comprehensive schedule format is specified elsewhere in the Contract Documents or voluntarily employed by the Contractor, ABC Form C-11 shall also be prepared, updated, and submitted as described in preceding Paragraph (b).

(d) The Contractor's construction schedule shall be used by the Contractor and Owner to determine the adequacy of the Contractor's progress. The Contractor shall be responsible for maintaining progress in accordance with the currently approved construction schedule and shall increase the number of shifts, and/or overtime operations, days of work, and/or the amount of construction plant and equipment as may be necessary to do so. If the Contractor's progress falls materially behind the currently approved construction schedule and, in the opinion of the Owner, the Contractor is not taking sufficient steps to regain schedule, the Owner may issue the Contractor a Notice to Cure pursuant to Section 3.27. In such a Notice to Cure the Owner may require the Contractor to submit such supplementary or revised construction schedules as may be deemed necessary to demonstrate the manner in which schedule will be regained.

3.13 EQUIPMENT, MATERIALS AND SUBSTITUTIONS.

(a) Every part of the Work shall be executed in a workmanlike manner in accordance with the Contract Documents and approved Submittals. All materials used in the Work shall be furnished in sufficient quantities to facilitate the proper and expeditious execution

of the Work and shall be new except such materials as may be expressly provided or allowed in the Contract Documents to be otherwise.

(b) Whenever a product, material, system, item of equipment, or service is identified in the Contract Documents by reference to a trade name, manufacturer's name, model number, etc. (hereinafter referred to as "source"), and only one or two sources are listed, or three or more sources are listed and followed by "or approved equal" or similar wording, it is intended to establish a required standard of performance, design, and quality, and the Contractor may submit, for the Owner's approval, products, materials, systems, equipment, or services of other sources which the Contractor can prove to the Owner's satisfaction are equal to, or exceed, the standard of performance, design and quality specified, unless the provisions of Paragraph (d) below apply. Such proposed substitutions are not to be purchased or installed without the Owner's written approval of the substitution.

(c) If the Contract Documents identify three or more sources for a product, material, system, item of equipment or service to be used and the list of sources is not followed by "or approved equal" or similar wording, the Contractor may make substitution only after evaluation by the Owner and execution of an appropriate Contract Change Order.

(d) If the Contract Documents identify only one source and expressly provide that it is an approved sole source for the product, material, system, item of equipment, or service, the Contractor must furnish the identified sole source.

3.14 SAFETY AND PROTECTION OF PERSONS AND PROPERTY.

(a) The Contractor shall be solely and completely responsible for conditions at the Project site, including safety of all persons (including employees) and property. The Contractor shall create, maintain, and supervise conditions and programs to facilitate and promote safe execution of the Work, and shall supervise the Work with the attention and skill required to assure its safe performance. Safety provisions shall conform to OSHA requirements and all other federal, state, county, and local laws, ordinances, codes, and regulations. Where any of these are in conflict, the more stringent requirement shall be followed.

(b) The Contractor shall employ Construction Methods, safety precautions, and protective measures that will reasonably prevent damage, injury or loss to:

- (i) Workers and other persons on the Project site and in adjacent and other areas that may be affected by the Contractor's operations;
- (ii) The Work and materials and equipment to be incorporated into the Work and stored by the Contractor on or off the Project site; and
- (iii) Other property on, or adjacent to, the Project site, including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities, and other improvements not designated in the Contract Documents to be removed, relocated, or replaced.

(c) The Contractor shall be responsible for the prompt remedy of damage and loss to property, including the filing of appropriate insurance claims, caused in whole or in part by the fault or negligence of the Contractor, a Subcontractor, or anyone for whose acts they may be liable.

(d) The Contractor shall comply with and give notices required by applicable laws, ordinances, rules, regulations, and lawful orders of public authorities bearing on safety and protection of persons or property, including without limitation notices to adjoining property

owners of excavation or other construction activities that potentially could cause damage or injury to adjoining property or persons thereon.

(e) The Contractor shall erect and maintain barriers, danger signs, and any other reasonable safeguards and warnings against hazards as may be required for safety and protection during performance of the Contract and shall notify owners and users of adjacent sites and utilities of conditions that may exist or arise which may jeopardize their safety.

(f) If use or storage of explosives or other hazardous materials or equipment or unusual construction methods are necessary for execution of the Work, the Contractor shall exercise commensurate care and employ supervisors and workers properly qualified to perform such activity.

(g) The Contractor shall furnish a qualified safety representative at the Project site whose duties shall include the prevention of accidents. The safety representative shall be the Contractor's superintendent, unless the Contractor assigns this duty to another responsible member of its on-site staff and notifies the Owner in writing of such assignment.

(h) The Contractor shall not permit a load to be applied, or forces introduced, to any part of the construction or site that may cause damage to the construction or site or endanger safety of the construction, site, or persons on or near the site.

(i) The Contractor shall have the right to act as it deems appropriate in emergency situations jeopardizing life or property. The Contractor shall be entitled to equitable adjustment of the Contract Sum or Contract Time for its efforts expended for the sole benefit of the Owner in an emergency. Such adjustment shall be determined as provided in Sections 3.19 and 3.20.

3.15 HAZARDOUS MATERIALS.

(a) A Hazardous Material is any substance or material identified as hazardous under any federal, state, or local law or regulation, or any other substance or material which may be considered hazardous or otherwise subject to statutory or regulatory requirements governing its handling, disposal, and/or clean-up. Existing Hazardous Materials are Hazardous Materials discovered at the Project site and not introduced to the Project site by the Contractor, a Subcontractor, or anyone for whose acts they may be liable.

(b) If during the performance of the Work, the Contractor encounters a suspected Existing Hazardous Material, the Contractor shall immediately stop work in the affected area, take measures appropriate to the condition to keep people away from the suspected Existing Hazardous Material, and immediately notify the Owner of the condition in writing.

(c) The Owner shall obtain the services of an independent laboratory or professional consultant, appropriately licensed and qualified, to determine whether the suspected material is a Hazardous Material requiring abatement and, if so, to certify after its abatement that it has been rendered harmless. Any abatement of Existing Hazardous Materials will be the responsibility of the Owner. The Owner will advise the Contractor in writing of the persons or entities who will determine the nature of the suspected material and those who will, if necessary, perform the abatement. The Owner will not employ persons or entities to perform these services to whom the Contractor has reasonable objection.

(d) After certification by the Owner's independent laboratory or professional consultant that the material is harmless or has been rendered harmless, work in the affected area shall resume upon written agreement between the Owner and Contractor. If the material is found to be an Existing Hazardous Material and the Contractor incurs additional cost or delay due to

the presence and abatement of the material, the Contract Sum and/or Contract Time shall be appropriately adjusted by a Contract Change Order pursuant to Section 3.19.

(e) The Owner shall not be responsible for Hazardous Materials introduced to the Project site by the Contractor, a Subcontractor, or anyone for whose acts they may be liable unless such Hazardous Materials were required by the Contract Documents.

3.16 INSPECTION OF THE WORK.

(a) General

- (i) The Contractor is solely responsible for the Work's compliance with the Contract Documents. Therefore, the Contractor shall be responsible to inspect in-progress and completed Work, and shall verify its compliance with the Contract Documents and that any element or portion of the Work upon which subsequent Work is to be applied or performed is in proper condition to receive the subsequent Work. Neither the presence nor absence of inspections by the Owner, any public authority having jurisdiction, or their representatives shall relieve the Contractor of responsibility to inspect the Work, for responsibility for Construction Methods and safety precautions and programs in connection with the Work, or from any other requirement of the Contract Documents.
- (ii) The Owner, any public authority having jurisdiction, and their representatives shall have access at all times to the Work for inspection whenever it is in preparation or progress, and the Contractor shall provide proper facilities for such access and inspection. All materials, workmanship, processes of manufacture, and methods of construction, if not otherwise stipulated in the Contract Documents, shall be subject to inspection, examination, and test at any and all places where such manufacture and/or construction are being carried on. Such inspections will not unreasonably interfere with the Contractor's operations.
- (iii) The Contractor may be charged by the Owner for any extra cost of inspection incurred by the Owner on account of material and workmanship not being ready at the time of inspection set by the Contractor.

(b) **Types of Inspections**

- (i) Scheduled Inspections and Conferences; Scheduled Inspections and Conferences are conducted by the Owner, scheduled by the Owner in coordination with the Contractor and are attended by the Contractor and applicable Subcontractors, suppliers and manufacturers. Scheduled Inspections and Conferences of this Contract include:
 - (1) **Pre-construction Conference**.
 - (2) **Final Inspection(s)**: A Final Inspection shall establish that the Work, or a designated portion of the Work, is Substantially Complete in accordance with Section 3.32 and is accepted by the Owner as being ready for the

Owner's occupancy or use. At the conclusion of this inspection, items requiring correction or completion ("punch list" items) shall be minimal and require only a short period of time for accomplishment to establish Final Acceptance of the Work. If the Work, or designated portion of the Work, includes the installation, or modification, of a fire alarm system or other life safety systems essential to occupancy, such systems shall have been tested and appropriately certified before the Final Inspection.

- (3) Year-end Inspection(s): An inspection of the Work or each separately completed portion thereof, is required near the end of the Contractor's one year warranty period(s). The subsequent delivery of the report of this inspection will serve as confirmation that the Contractor was notified of Defective Work found within the warranty period in accordance with Section 3.35.
- (ii) **Periodic Inspections**. Periodic Inspections are conducted throughout the course of the Work by the Owner, the Owner's consultants, and their representatives, jointly or independently, with or without advance notice to the Contractor.
- (iii) Specified Inspections and Tests. Specified Inspections and Tests include inspections, tests, demonstrations, and approvals that are either specified in the Contract Documents or required by laws, ordinances, rules, regulations, or orders of public authorities having jurisdiction, to be performed by the Contractor, one of its Subcontractors, or an independent testing laboratory or firm (whether paid for by the Contractor or Owner).

(c) Uncovering Work.

- (i) If the Contractor covers a portion of the Work before it is examined by the Inspector and this is contrary to the Owner's request or specific requirements in the Contract Documents, then, upon written request of the Owner, the Work must be uncovered for the Inspector's examination and be replaced at the Contractor's expense without change in the Contract Time.
- (ii) Without a prior request or specific requirement that Work be examined by the Inspector before it is covered, the Inspector may request that Work be uncovered for examination and the Contractor shall uncover it. If the Work is in accordance with the Contract Documents, the Contract Sum shall be equitably adjusted under Section 3.19 to compensate the Contractor for the costs of uncovering and replacement. If the Work is not in accordance with the Contract Documents, uncovering, correction, and replacement shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

(d) Specified Inspections and Tests.

- The Contractor shall schedule and coordinate Specified Inspections (i) and Tests to be made at appropriate times so as not to delay the progress of the Work or the work of the Owner or separate contractors. If the Contract Documents require that a Specified Inspection or Test be witnessed or attended by the Owner's Representative, the Contractor shall give the Owner timely notice of the time and place of the Specified Inspection or Test. If a Specified Inspection or Test reveals that Work is not in compliance with requirements of the Contract Documents, the Contractor shall bear the costs of correction, repeating the Specified Inspection or Test, and any related costs incurred by the Owner. Through appropriate Contract Change Order the Owner shall bear costs of inspections or approvals which become Contract tests. requirements subsequent to the receipt of bids.
- (ii) If the Owner or public authority having jurisdiction determines that inspections, tests, demonstrations, or approvals in addition to Specified Inspections and Tests are required, the Contractor shall, upon written instruction from the Owner, arrange for their performance by an entity acceptable to the Owner, giving timely notice to the Owner of the time and place of their performance. Related costs shall be borne by the Owner unless the procedures reveal that Work is not in compliance with requirements of the Contract Documents, in which case the Contractor shall bear the costs of correction, repeating the procedures, and any related costs incurred by the Owner.
- (iii) Unless otherwise required by the Contract Documents, required certificates of Specified Inspections and Tests shall be secured by the Contractor and promptly delivered to the Owner.
- (iv) Failure of any materials to pass Specified Inspections and Tests will be sufficient cause for refusal to consider any further samples of the same brand or make of that material for use in the Work.

3.17 CORRECTION OF DEFECTIVE WORK.

(a) The Contractor shall, at the Contractor's expense, promptly correct Defective Work rejected by the Owner or which otherwise becomes known to the Contractor, removing the rejected or nonconforming materials and construction from the project site.

(b) Correction of Defective Work shall be performed in such a timely manner as will avoid delay of completion, use, or occupancy of the Work and the work of the Owner and separate contractors.

(c) The Contractor shall bear all expenses related to the correction of Defective Work, including but not limited to:

- (i) Additional testing and inspections, including repeating specified inspections and tests,
- (ii) Reasonable services and expenses of the Owner, and

(iii) The expense of making good all work of the Contractor, Owner, or separate contractors destroyed or damaged by the correction of Defective Work.

3.18 DEDUCTIONS FOR UNCORRECTED WORK. If the Owner deems it advisable and in the Owner's interest to accept Defective Work, the Owner may allow part or all of such Work to remain in place, provided an equitable deduction from the Contract Sum, acceptable to the Owner, is offered by the Contractor.

3.19 CHANGES IN THE WORK.

(a) General.

- (i) The Owner may at any time direct the Contractor to make changes in the Work which are within the general scope of the Contract, including changes in the Drawings, Specifications, or other portions of the Contract Documents to add, delete, or otherwise revise portions of the Work.
- (ii) If the Owner directs a change in the Work, the change shall be incorporated into the Contract by a Contract Change Order prepared by the Owner and signed by the Contractor, Owner, and other signatories to the Construction Contract, stating their agreement upon the change or changes in the Work and the adjustments, if any, in the Contract Sum and the Contract Time.
- (iii) In the event of a claim or dispute as to the appropriate adjustment to the Contract Sum or Contract Time due to a directive to make changes in the Work, the Work shall proceed as provided in this section subject to subsequent agreement of the parties or final resolution of the dispute pursuant to Section 3.24.
- (iv) Consent of surety will be obtained from all Contract Change Orders involving an increase in the Contract Sum.
- (v) Changes in the Work shall be performed under applicable provisions of the Contract Documents and the Contractor shall proceed promptly to perform changes in the Work, unless otherwise directed by the Owner.
- (b) **Determination of Adjustment of the Contract Sum**. The adjustment of the Contract Sum resulting from a change in the Work shall be determined by one of the following methods, or a combination thereof, as selected by the Owner:
 - Lump Sum. By mutual agreement to a lump sum based on or negotiated from an itemized cost proposal from the Contractor. Additions to the Contract Sum shall include the Contractor's direct costs plus a maximum 15% markup for overhead and profit. Where subcontract work is involved the total mark-up for the Contractor and a Subcontractor shall not exceed 25%. No allowance for overhead and profit shall be figured on a change which involves a net credit to the Owner. For the purposes of this method of determining an adjustment of the Contract Sum,

"overhead" shall cover the Contractor's indirect costs of the change, such as the cost of bonds, superintendent and other job office personnel, watchman, job office, job office supplies and expenses, temporary facilities and utilities, and home office expenses.

- (ii) Unit Price. By application of Unit Prices included in the Contract or subsequently agreed to by the parties. However, if the character or quantity originally contemplated is materially changed so that application of such unit price to quantities of Work proposed will cause substantial inequity to either party, the applicable unit price shall be equitably adjusted.
- (iii) Force Account. By directing the Contractor to proceed with the change in the Work on a "force account" basis under which the Contractor shall be reimbursed for reasonable expenditures incurred by the Contractor and its Subcontractors in performing added Work and the Owner shall receive reasonable credit for any deleted Work. The Contractor shall keep and present, in such form as the Owner may prescribe, an itemized accounting of the cost of the change together with sufficient supporting data. Unless otherwise stated in the directive, the adjustment of the Contract Sum shall be limited to the following:
 - (1) Costs of labor and supervision, including employee benefits, social security, retirement, unemployment and workers' compensation insurance required by law, agreement, or under Contractor's or Subcontractor's standard personnel policy;
 - (2) Cost of materials, supplies, and equipment, including cost of delivery, whether incorporated or consumed;
 - (3) Rental cost of machinery and equipment, not to exceed prevailing local rates if contractor-owned;
 - (4) Costs of premiums for insurance required by the Contract Documents, permit fees, and sales, use or similar taxes related to the change in the Work;
 - (5) Reasonable credits to the Owner for the value of deleted Work, without Contractor or Subcontractor mark-ups; and
 - (6) For additions to the Contract Sum, mark-up of the Contractor's direct costs for overhead and profit not exceeding 15% on Contractor's work nor exceeding 25% for Contractor and Subcontractor on a Subcontractor's work. No allowance for overhead and profit shall be figured on a change which involves a net credit to the Owner. For the purposes of this method of determining an adjustment of the Contract Sum, "overhead" shall cover the Contractor's indirect costs of the change, such as the cost of insurance other than mentioned above, bonds, superintendent and other job office personnel, watchman,

use and rental of small tools, job office, job office supplies and expenses, temporary facilities and utilities, and home office expenses.

(c) Adjustment of the Contract Time Due to Changes.

- (i) Unless otherwise provided in the Contract Documents, the Contract Time shall be equitably adjusted for the performance of a change provided that the Contractor notifies the Owner in writing that the change will increase the time required to complete the Work. Such notice shall be provided no later than:
 - (1) With the Contractor's cost proposal stating the number of days of extension requested, or
 - (2) Within ten days after the Contractor receives a directive to proceed with a change in advance of submitting a cost proposal, in which case the notice should provide an estimated number of days of extension to be requested, which may be subject to adjustment in the cost proposal.

The Contract Time shall be extended only to the extent that the change affects the time required to complete the entire Work of the Contract, taking into account the concurrent performance of the changed and unchanged Work.

(d) Change Order Procedures.

- (i) If the Owner proposes to make a change in the Work, the Owner will request that the Contractor provide a cost proposal for making the change to the Work. The request shall be in writing and shall adequately describe the proposed change using drawings, specifications, narrative, or a combination thereof. Within 21 days after receiving such a request, or such other time as may be stated in the request, the Contractor shall prepare and submit to the Owner a written proposal, properly itemized and supported by sufficient substantiating data to facilitate evaluation. The stated time within which the Contractor must submit a proposal may be extended if, within that time, the Contractor makes a written request with reasonable justification thereof.
- (ii) The Contractor may voluntarily offer a change proposal which, in the Contractor's opinion, will reduce the cost of construction, maintenance, or operation or will improve the cost-effective performance of an element of the Project, in which case the Owner will accept, reject, or respond otherwise within 21 days after receipt of the proposal, or such other reasonable time as the Contractor may state in the proposal.
- (iii) If the Contractor's proposal is acceptable to the Owner, or is negotiated to the mutual agreement of the Contractor and Owner, the Owner will prepare an appropriate Contract Change Order for execution. Upon receipt of the fully executed Contract Change Order, the Contractor shall proceed with the change.

- (iv) If the Contractor and Owner cannot agree on the amount of the adjustment in the Contract Sum for a change, the Owner may order the Contractor to proceed with the change on a Force Account basis, but the net cost to the Owner shall not exceed the amount quoted in the Contractor's proposal. Such order shall state that funds are available to pay for the change.
- (v) If the Contractor does not promptly respond to a request for a proposal, or the Owner determines that the change is essential to the final product of the Work and that the change must be effected immediately to avoid delay of the Project, the Owner may determine with the Contractor a sufficient maximum amount to be authorized for the change and direct the Contractor to proceed with the change on a Force Account basis pending delivery of the Contractor's proposal, stating the maximum increase in the Contract Sum that is authorized for the change.
- (iv) Pending agreement of the parties or final resolution of any dispute of the total amount due the Contractor for a change in the Work, amounts not in dispute for such changes in the Work may be included in Applications for Payment accompanied by an interim Change Order indicating the parties' agreement with part of all of such costs or time extension. Once a dispute is resolved, it shall be implemented by preparation and execution of an appropriate Change Order.

3.20 CLAIMS FOR EXTRA COST OR EXTRA WORK.

(a) If the Contractor considers any instructions by the Owner, or public authority having jurisdiction to be contrary to the requirements of the Contract Documents and will involve extra work and/or cost under the Contract, the Contractor shall give the Owner written notice thereof within ten days after receipt of such instructions, and in any event before proceeding to execute such work. As used in this section, "instructions" shall include written or oral clarifications, directions, instructions, interpretations, or determinations.

- The Contractor's notification pursuant to Section 3.20. (a), shall state:
 - (i) The date, circumstances, and source of the instructions,
 - (ii) That the Contractor considers the instructions to constitute a change to the Contract Documents and why, and
 - (iii) An estimate of extra cost and time that may be involved to the extent an estimate may be reasonably made at that time.

(c) Except for claims relating to an emergency endangering life or property, no claim for extra cost or extra work shall be considered in the absence of prior notice required under Section 3.20 (a).

(d) Within ten days of receipt of a notice pursuant to Section 3.20 (a), the Owner will respond in writing to the Contractor, stating one of the following:

- (i) The cited instruction is rescinded.
- (ii) The cited instruction is a change in the Work and in which manner the Contractor is to proceed with procedures of Section 3.19, Changes in the Work.

(b)

(iii) The cited instruction is reconfirmed, is not considered by the Owner to be a change in the Contract Documents, and the Contractor is to proceed with Work as instructed.

(e) If the Owner's response to the Contractor is as in Section 3.20 (d) (iii), the Contractor shall proceed with the Work as instructed. If the Contractor continues to consider the instructions to constitute a change in the Contract Documents, the Contractor shall, within ten days after receiving the Owner's response, notify the Owner in writing that the Contractor intends to submit a claim pursuant to Section 3.24, Resolution of Claims and Disputes.

3.21 DIFFERING SITE CONDITIONS.

(a) **Definition.** "Differing Site Conditions" are: Subsurface or otherwise concealed physical conditions at the Project site which differ materially from those indicated in the Contract Documents, or unknown physical conditions at the Project site which are of an unusual nature, differing materially from conditions ordinarily encountered and generally recognized as inherent in construction activities of the character required by the Contract Documents.

(b) **Procedures**. If Differing Site Conditions are encountered, then the party discovering the condition shall promptly notify the other party before the condition is disturbed and in no event later than ten days after discovering the condition. Upon such notice and verification that a Differing Site Condition exists, the Owner will, with reasonable promptness make changes in the Drawings and/or Specifications as are deemed necessary to conform to the Differing Site Condition. Any increase or decrease in the Contract Sum or Contract Time that is warranted by the changes will be made as provided under Section 3.19, Changes in the Work.

3.22 CLAIMS FOR DAMAGES. If either party to the Contract suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time after the discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

3.23 DELAYS.

(a) A delay beyond the Contractor's control at any time in the commencement or progress of Work by an act or omission of the Owner, or any separate contractor or by labor disputes, unusual delay in deliveries, unavoidable casualties, fires, abnormal floods, tornadoes, or other cataclysmic events of nature, may entitle the Contractor to an extension of the Contract Time provided, however, that the Contractor shall, within ten days after the delay first occurs, give written notice to the Owner of the cause of the delay and its probable effect on progress of the entire Work.

(b) Adverse weather conditions that are more severe than anticipated for the locality of the Work during any given month may entitle the Contractor to an extension of Contract Time provided, however;

(i) The weather conditions had an adverse effect on construction scheduled to be performed during the period in which the adverse weather occurred, which in reasonable sequence would have an effect on completion of the entire Work,

- (ii) The Contractor shall, within twenty-one days after the end of the month in which the delay occurs, give the Owner written notice of the delay that occurred during that month and its probable effect on progress of the Work, and
- (iii) Within a reasonable time after giving notice of the delay, the Contractor provides the Owner with sufficient data to document that the weather conditions experienced were unusually severe for the locality of the Work during the month in question. Unless otherwise provided in the Contract Documents, data documenting unusually severe weather conditions shall compare actual weather conditions to the average weather conditions for the month in question during the previous five years as recorded by the National Oceanic and Atmospheric Administration (NOAA) or similar record-keeping entities.

(c) Adjustments, if any, of the Contract Time pursuant to this Section shall be incorporated into the Contract by a Contract Change Order prepared by the Owner and signed by the Contractor, Owner, and other signatories to the Construction Contract or, at closeout of the Contract, by mutual written agreement between the Contractor and Owner. The adjustment of the Contract Time shall not exceed the extent to which the delay extends the time required to complete the entire Work of the Contract.

(d) The Contractor shall not be entitled to any adjustment of the Contract Sum for damage due to delays claimed pursuant to this Section unless the delay was caused by the Owner and was either:

- (i) The result of bad faith or active interference or
- (ii) Beyond the contemplation of the parties and not remedied within a reasonable time after notification by the Contractor of its presence.

3.24 RESOLUTION OF CLAIMS AND DISPUTES.

(a) Applicability of Section.

- (i) As used in this section, "Claims and Disputes" include claims or disputes asserted by the Contractor, its Surety, or Owner arising out of or related to the Contract, or its breach, including without limitation claims seeking, under the provisions of the Contract, equitable adjustment of the Contract Sum or Contract Time and claims and disputes arising between the Contractor (or its Surety) and Owner regarding interpretation of the Contract Documents, performance of the Work, or breach of or compliance with the terms of the Contract.
- (ii) "Resolution" addressed in this Section applies only to Claims and Disputes arising between the Contractor (and its Surety) and Owner and asserted after execution of the Construction Contract and prior to the date upon which final payment is made. Upon making application for final payment the Contractor may reserve the right to subsequent Resolution of existing Claims by including a list of all Claims, in stated amounts, which remain to be resolved and specifically excluding them from any release of claims

executed by the Contractor, and in that event Resolution may occur after final payment is made.

(b) Continuance of Performance. An unresolved Claim or Dispute shall not be just cause for the Contractor to fail or refuse to proceed diligently with performance of the Contract or for the Owner to fail or refuse to continue to make payments in accordance with the Contract Documents.

(c) Good Faith Effort to Settle. The Contractor and Owner agree that, upon the assertion of a Claim by the other, they will make a good faith effort to achieve mutual resolution of the Claim. If mutually agreed, the Contractor and Owner may endeavor to resolve a Claim through mediation. If efforts to settle are not successful, the Claim shall be resolved in accordance with paragraph (d).

(d) Final Resolution for Locally-Funded Contracts. If the Contract is funded in whole with funds provided by a local governmental authority and the Contract Documents do not stipulate a binding alternative dispute resolution method, the final resolution of Claims and Disputes which cannot be resolved by the Contractor (or its Surety) and Owner may be by any legal remedy available to the parties. Alternatively, upon the written agreement of the Contractor and the Owner, final Resolution of Claims and Disputes may be by submission to binding arbitration before a neutral arbitrator or panel.

3.25 OWNER'S RIGHT TO CORRECT DEFECTIVE WORK. If the Contractor fails or refuses to correct Defective Work in a timely manner that will avoid delay of completion, use, or occupancy of the Work or work by the Owner or separate contractors, the Owner may give the Contractor written Notice to Cure the Defective Work within a reasonable, stated time. If within ten days after receipt of the Notice to Cure the Contractor has not proceeded and satisfactorily continued to cure the Defective Work or provided the Owner with written verification that satisfactory positive action is in process to cure the Defective Work, the Owner may, without prejudice to any other remedy available to the Owner, correct the Defective Work and deduct the actual cost of the correction from payment then or thereafter due to the Contractor.

3.26 OWNER'S RIGHT TO STOP OR SUSPEND THE WORK.

(a) **Stopping the Work for Cause**. If the Contractor fails to correct Defective Work or persistently fails to carry out Work in accordance with the Contract Documents, the Owner may direct the Contractor in writing to stop the Work, or any part of the Work, until the cause for the Owner's directive has been eliminated; however, the Owner's right to stop the Work shall not be construed as a duty of the Owner to be exercised for the benefit of the Contractor or any other person or entity.

(b) Suspension by the Owner for Convenience.

- (i) The Owner may, at any time and without cause, direct the Contractor in writing to suspend, delay or interrupt the Work, or any part of the Work, for a period of time as the Owner may determine.
- (ii) The Contract Sum and Contract Time shall be adjusted, pursuant to Section 3.19, for reasonable increases in the cost and time caused by an Owner-directed suspension, delay or interruption of Work for the Owner's convenience. However, no adjustment to the

Contract Sum shall be made to the extent that the same or concurrent Work is, was or would have been likewise suspended, delayed or interrupted for other reasons not caused by the Owner.

3.27 OWNER'S RIGHT TO TERMINATE CONTRACT.

(a) Termination by the Owner for Cause.

- (i) **Causes**: The Owner may terminate the Contractor's right to complete the Work, or any designated portion of the Work, if the Contractor:
 - (1) Should be adjudged bankrupt, or should make a general assignment for the benefit of the Contractor's creditors, or if a receiver should be appointed on account of the Contractor's insolvency to the extent termination for these reasons is permissible under applicable law;
 - (2) Refuses or fails to prosecute the Work, or any part of the Work, with the diligence that will insure its completion within the Contract Time, including any extensions, or fails to complete the Work within the Contract Time;
 - (3) Refuses or fails to perform the Work, including prompt correction of Defective Work, in a manner that will insure that the Work, when fully completed, will be in accordance with the Contract Documents;
 - (4) Fails to pay for labor or materials supplied for the Work or to pay Subcontractors in accordance with the respective Subcontract;
 - (5) Persistently disregards laws, ordinances, or rules, regulations or orders of a public authority having jurisdiction, or the instructions of the Owner; or
 - (6) Is otherwise guilty of a substantial breach of the Contract.
- (ii) **Procedure for Unbonded Construction Contracts** (Generally, contracts less than \$50,000):
 - (1) **Notice to Cure**: In the presence of any of the above conditions the Owner may give the Contractor written notice to cure the condition within a reasonable, stated time, but not less than ten days after the Contractor receives the notice.
 - (2) **Notice of Termination**: If, at the expiration of the time stated in the Notice to Cure, the Contractor has not proceeded and satisfactorily continued to cure the condition or provided the Owner with written verification that satisfactory positive action is in process to cure the condition, the Owner may, without prejudice to any other rights or remedies of the Owner, give the Contractor written notice that the Contractor's right to complete the Work, or a designated portion of the Work, shall terminate

seven days after the Contractor's receipt of the written Notice of Termination.

- (3) If the Contractor satisfies a Notice to Cure, but the condition for which the notice was first given reoccurs, the Owner may give the Contractor a seven day Notice of Termination without giving the Contractor another Notice to Cure.
- (4) At the expiration of the seven days of the termination notice, the Owner may:

a. take possession of the site, of all materials and equipment stored on and off site, and of all Contractorowned tools, construction equipment and machinery, and facilities located at the site, and

b. finishes the Work by whatever reasonable method the Owner may deem expedient.

- (5) The Contractor shall not be entitled to receive further payment under the Contract until the Work is completed.
- (6) If the Owner's cost of completing the Work, including correction of Defective Work, compensation for additional architectural, engineering, managerial, and administrative services, and reasonable attorneys' fees due to the default and termination, is less than the unpaid balance of the Contract Sum, the excess balance less liquidated damages for delay shall be paid to the Contractor. If such cost to the Owner including attorney's fees, plus liquidated damages, exceeds the unpaid balance of the Contract Sum, the Contract Sum, the difference to the Owner. Final Resolution of any claim or Dispute involving the termination or any amount due any party as a result of the termination shall be pursuant to Section 3.24.
- (7) Upon the Contractor's request, the Owner shall furnish to the Contractor a detailed accounting of the Owner's cost of completing the Work.
- (iii) **Procedure for Bonded Construction Contracts** (Generally, contracts over \$50,000):
 - (1) **Notice to Cure:** In the presence of any of the above conditions the Owner may give the Contractor and its Surety written Notice to Cure the condition within a reasonable, stated time, but not less than ten days after the Contractor receives the notice.
 - (2) **Notice of Termination:** If, at the expiration of the time stated in the Notice to Cure, the Contractor has not proceeded and satisfactorily continued to cure the condition or provided the Owner with written verification that satisfactory positive action is in process to cure the condition, the Owner may, without prejudice to any other

rights or remedies of the Owner, give the Contractor and its Surety written notice declaring the Contractor to be in default under the Contract and stating that the Contractor's right to complete the Work, or a designated portion of the Work, shall terminate seven days after the Contractor's receipt of the written Notice of Termination.

- (3) If the Contractor satisfies a Notice to cure, but the condition for which the notice was first given reoccurs, the Owner may give the Contractor a Notice of Termination without giving the Contractor another Notice to Cure.
- (4) **Demand on the Performance Bond**: With the Notice of Termination the Owner shall give the Surety a written demand that, upon the effective date of the Notice of Termination, the Surety promptly fulfill its obligation to take charge of and complete the Work in accordance with the terms of the Performance Bond.
- Surety Claims: Upon receiving the Owner's demand on (5) the Performance Bond, the Surety shall assume all rights and obligations of the Contractor under the Contract. However, the Surety shall also have the right to assert "Surety Claims" to the Owner, which are defined as claims relating to acts or omissions of the Owner prior to termination of the Contractor which may have prejudiced its rights as Surety or its interest in the unpaid balance of the Contract Sum. If the Surety wishes to assert a Surety Claim, it shall give the Owner written notice within twentyone days after first recognizing the condition giving rise to The Surety Claim shall then be the Surety Claim. submitted to the Owner no later than sixty days after giving notice thereof, but no such Surety Claims shall be considered if submitted after the date upon which final payment becomes due. Final resolution of Surety Claims shall be pursuant to Section 3.24, Resolution of Claims and Disputes. The presence or possibility of a Surety Claim shall not be just cause for the Surety to fail or refuse to take charge of and complete the Work or for the Owner to fail or refuse to continue to make payments in accordance with the Contract Documents.
- (6) **Payments to Surety**: The Surety shall be paid for completing the Work in accordance with the Contract Documents as if the Surety were the Contractor. The Owner shall have the right to deduct from payments to the Surety any reasonable costs incurred by the Owner, including compensation for additional architectural, engineering, managerial, and administrative services, and attorneys' fees as necessitated by termination of the

Contractor and completion of the Work by the Surety. No further payments shall be made to the Contractor by the Owner. The Surety shall be solely responsible for any accounting to the Contractor for the portion of the Contract Sum paid to Surety by Owner or for the costs and expenses of completing the Work.

(iv) Wrongful Termination: If any notice of termination by the Owner for cause, made in good faith, is determined to have been wrongly given, such termination shall be effective and compensation therefore determined as if it had been a termination for convenience pursuant to Paragraph (b) below.

(b) Termination by the Owner for Convenience.

- (i) The Owner may, without cause and at any time, terminate the performance of Work under the Contract in whole, or in part, upon determination by the Owner that such termination is in the Owner's best interest. Such termination is referred to herein as Termination for Convenience.
- (ii) Upon receipt of a written notice of Termination for Convenience from the Owner, the Contractor shall:
 - (1) Stop Work as specified in the notice;
 - (2) Enter into no further subcontracts or purchase orders for materials, services, or facilities, except as may be necessary for Work directed to be performed prior to the effective date of the termination or to complete Work that is not terminated;
 - (3) Terminate all existing subcontracts and purchase orders to the extent they relate to the terminated Work;
 - (4) Take such actions as are necessary, or directed by the Owner, to protect, preserve, and make safe the terminated Work; and
 - (5) Complete performance of the Work that is not terminated.
- (iii) In the event of Termination for Convenience, the Contractor shall be entitled to receive payment for the Work performed prior to its termination, including materials and equipment purchased and delivered for incorporation into the terminated Work, and any reasonable costs incurred because of the termination. Such payment shall include reasonable mark-up of costs for overhead and profit, not to exceed the limits stated in Section 3.19, Changes in the Work. The Contractor shall be entitled to receive payment for reasonable anticipated overhead ("home office") and shall not be entitled to receive payment for any profits anticipated to have been gained from the terminated Work. A proposal for decreasing the Contract Sum shall be submitted to the Owner by the Contractor in such time and detail, and with such supporting documentation, as is reasonably directed by the Owner. Final modification of the Contract shall be by Contract Change Order

pursuant to Section 3.19. Any Claim or Dispute involving the termination or any amount due a party as a result shall be resolved pursuant to Section 3.24.

3.28 CONTACTOR'S RIGHT TO SUSPEND OR TERMINATE THE CONTRACT.

(a) **Suspension by the Owner**. If all of the Work is suspended or delayed for the Owner's convenience or under an order of any court, or other public authority, for a period of sixty days, through no act or fault of the Contractor or a Subcontractor, or anyone for whose acts they may be liable, then the Contractor may give the Owner a written Notice of Termination which allows the Owner fourteen days after receiving the Notice in which to give the Contractor appropriate written authorization to resume the Work. Absent the Contractor's receipt of such authorization to resume the Work, the Contract shall terminate upon expiration of this fourteen day period and the Contractor will be compensated by the Owner as if the termination had been for the Owner's convenience pursuant to Section 3.27 (b).

(b) Nonpayment. The Owner's failure to pay the undisputed amount of an Application for Payment within sixty days after receiving it from the Owner shall be just cause for the Contractor to give the Owner fourteen days' written notice that the Work will be suspended pending receipt of payment but that the Contract shall terminate if payment is not received within fourteen days (or a longer period stated by the Contractor) of the expiration of the fourteen day notice period.

- (i) If the Work is then suspended for nonpayment, but resumed upon receipt of payment, the Contractor will be entitled to compensation as if the suspension had been by the Owner pursuant to Section 3.26, paragraph (b).
- (ii) If the Contract is then terminated for nonpayment, the Contractor will be entitled to compensation as if the termination had been by the Owner pursuant to Section 3.27, Paragraph (b).

3.29 PROGRESS PAYMENTS.

(a) **Frequency of Progress Payments**. Unless otherwise provided in the Contract Documents, the Owner will make payments to the Contractor as the Work progresses based on estimates prepared and certified by the Contractor at the end of each calendar month and approved by the Owner and other authorities whose approval is required.

(b) **Designation of Owner's Representative**. All progress payments and associated documentation shall be submitted to the Owner's Representative for review, approval, and certification as per Section 3.30.

(c) Schedule Values. Within ten days after receiving the Notice to Proceed the Contractor shall submit to the Owner a Schedule of Values, which is a breakdown of the Contract Sum showing the value of the various parts of the Work for billing purposes, in a format acceptable to the Owner. The Schedule of Values shall divide the Contract Sum into as many parts ("line items") as the Owner determines to be necessary to permit evaluation and to show amounts attributable to Subcontractors. The Contractor's overhead and profit are to be proportionately distributed throughout the line items of the Schedule of Values. Upon approval, the Schedule of Values shall be used as a basis for monthly Applications for Payment, unless it is

later found to be in error. Approved change order amounts shall be added to or incorporated into the Schedule of Values as mutually agreed by the Contractor and Owner.

(d) Application for Payments.

- Based on the approved Schedule of Values, each monthly Application for Payment shall show the Contractor's estimate of the value of Work performed in each line item as of the end of the billing period. The Contractor's cost of materials and equipment not yet incorporated into the Work, but delivered and suitably stored on the site, may be considered in monthly Applications for Payment.
- (ii) The Contractor's estimate of the value of Work performed and stored materials must represent such reasonableness as to warrant certification by the Owner in accordance with Section 3.30. Each monthly Application for Payment shall be supported by such data as will substantiate the Contractor's right to payment, including without limitation copies of requisitions from subcontractors and material suppliers.
- (iii) If no other date is stated in the Contract Documents or agreed upon by the parties, each monthly Application for Payment shall be submitted to the Owner on or about the first day of each month and payment shall be issued to the Contractor within thirty five (35) days after an Application for Payment is certified pursuant to Section 3.30.

(e) Materials Stored Off Site. Unless otherwise provided in the Contract Documents, the Contractor's cost of materials and equipment to be incorporated into the Work, which are stored off the site, may also be considered in monthly Applications for Payment under the following conditions:

- (i) The contractor has received written approval from the Owner to store the materials or equipment off site in advance of delivering the materials to the off site location;
- (ii) A Certificate of Insurance is furnished to the Owner evidencing that a special insurance policy, or rider to an existing policy, has been obtained by the Contractor providing all-risk property insurance coverage, specifically naming the materials or equipment stored, and naming the Owner as an additionally insured party;
- (iii) The Owner is provided with a detailed inventory of the stored materials or equipment and the materials or equipment are clearly marked in correlation to the inventory to facilitate inspection and verification of the presence of the materials or equipment by the Owner;
- (iv) The materials or equipment are properly and safely stored in a bonded warehouse, or a facility otherwise approved in advance by the Owner; and
- (v) Compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest.

(f) Retainage.

- (i) "Retainage" is defined as the money earned and, therefore, belonging to the Contractor (subject to final settlement of the Contract) which has been retained by the Owner conditioned on final completion and acceptance of all Work required by the Contract Documents. Retainage shall not be relied upon by Contractor (or Surety) to cover or off-set unearned monies attributable to uncompleted or uncorrected Work.
- (ii) In making progress payments the Owner shall retain five percent of the estimated value of Work performed and the value of the materials stored for the Work; but after retainage has been held upon fifty percent of the Contract Sum, no additional retainage will be withheld.

(g) Contractor's Certification.

- (i) Each Application for Payment shall bear the Contractor's notarized certification that, to the best of the Contractor's knowledge, information, and belief, the Work covered by the Application for Payment has been completed in accordance with the Contract Documents, that all amounts have been paid by the Contractor for Work for which previous Certificates for Payments were issued and payments received from the Owner and that the current payment shown in the Application for Payment has not yet been received.
- (ii) By making this certification the Contractor represents to the Owner that, upon receipt of previous progress payments from the Owner, the Contractor has promptly paid each Subcontractor, in accordance with the terms of its agreement with the Subcontractor, the amount due the Subcontractor from the amount included in the progress payment on account of the Subcontractor's Work and stored materials. The Owner may advise Subcontractors and suppliers regarding percentages of completion or amounts requested and/or approved in an Application for Payment on account of the Subcontractor's Work and stored materials.

(h) **Payment Established Ownership**. All material and Work covered by progress payments shall become the sole property of the Owner, but the Contractor shall not be relieved from the sole responsibility for the care and protection of material and Work upon which payments have been made and for the restoration of any damaged material and Work.

3.30 CERTIFICATION AND APPROVALS FOR PAYMENT.

(a) The Owner's review, approval, and certification of Applications for Payment shall be conducted by the Owner's Representative. The Owner's Representative's review, approval, and certification of Applications for Payment shall be based on the Owner's Representative's general knowledge of the Work obtained through site visits and the information provided by the Contractor with the Application. The Owner's Representative shall not be required to perform exhaustive examinations, evaluations, or estimates of the cost of completed or uncompleted Work or stored materials to verify the accuracy of amounts requested by the Contractor, but the Owner's Representative shall have the authority to adjust the Contractor's estimate when, in the Owner's Representative's reasonable opinion, such estimates are overstated or understated.

(b) Within ten (10) days after receiving the Contractor's monthly Application for Payment, or such other time as may be stated in the Contract Documents, the Owner's Representative will take one of the following actions:

- (i) The Owner's Representative will approve and certify the Application as submitted and forward for payment.
- (ii) If the Owner's Representative takes exception to any amounts claimed by the Contractor and the Contractor and Owner's Representative cannot agree on revised amounts, the Owner's representative will promptly issue a Certificate for Payment for the amount for which the Owner's Representative is able to certify, transmitting a copy of same to the Contractor.
- (iii) To the extent the Owner's Representative determines may be necessary to protect from loss on account of any of the causes stated in Section 3.31, the Owner's Representative may subtract from the Contractor's estimates and will issue a Certificate for Payment with a copy to the Contractor, for such amount as the Owner's Representative determines is properly due and notify the Contractor in writing of the reasons for withholding payment in whole or in part.

(c) Neither the Owner's Representative's issuance of a Certificate for Payment nor the Owner's resulting progress payment shall be a representation to the Contractor that the Work in progress or completed at that time is accepted or deemed to be in conformance with the Contract Documents.

(d) The Owner's Representative shall not be required to determine that the Contractor has promptly or fully paid Subcontractors and suppliers or how or for what purpose the Contractor has used monies paid under the Construction Contract. However, the Owner's Representative may, upon request and if practical, inform any Subcontractor or supplier of the amount, or percentage of completion, approved or paid to the Contractor on account of the materials supplied or the Work performed by the Subcontract.

3.31 PAYMENTS WITHHELD.

(a) The Owner may nullify or revise a previously issued Certificate for payment prior to Owner's payment there under to the extent as may be necessary in the Owner's opinion to protect the Owner from loss on account of any of the following causes not discovered or fully accounted for at the time of the certification or approval of the Application for Payment:

- (i) Defective Work;
- (ii) Filed, or reasonable evidence indicating probable filing of, claims arising out of the Contract by other parties against the Contractor;
- (iii) The Contractor's failure to pay for labor, materials or equipment or to pay Subcontractors;
- (iv) Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;

- (v) Damage suffered by the Owner or another contractor caused by the Contractor, a Subcontractor, or anyone for whose acts they may be liable;
- (vi) Reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance is insufficient to cover applicable liquidated damages; or
- (vii) The Contractor's persistent failure to conform to the requirements of the Contract Documents.

(b) If the Owner deems it necessary to withhold payment pursuant to preceding Paragraph (a), the Owner will notify the Contractor in writing of the amount to be withheld and the reason for same.

(c) The Owner shall not be required to withhold payment for completed or partially completed Work for which compliance with the Contract Documents remains to be determined by Specified Inspections or Final Inspections to be performed in their proper sequence. However, if Work for which payment has been approved, certified, or made under an Application for Payment is subsequently determined to be Defective Work, the Owner shall determine an appropriate amount that will protect the Owner's interest against the Defective Work.

- (i) If payment has not been made against the Application for Payment first including the Defective Work, the Owner will notify the Contractor of the amount to be withheld from the payment until the Defective Work is brought into compliance with the Contract Documents.
- (ii) If payment has been made against the Application for Payment first including the Defective Work, the Owner will withhold the appropriate amount from the next Application for Payment submitted after the determination of noncompliance, such amount to then be withheld until the Defective Work is brought into compliance with the Contract Documents.

(d) The amount withheld will be paid with the next Application for Payment certified and approved after the condition for which the Owner has withheld payment is removed or otherwise resolved to the Owner's satisfaction.

(e) The Owner shall have the right to withhold from payments due the Contractor under this Contract an amount equal to any amount which the Contractor owes the Owner under another contract.

3.32 SUBSTANTIAL COMPLETION.

(a) Substantial Completion is the stage in the progress of the Work when the Work or designated portion of the Work is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use without disruption or interference by the Contractor in completing or correcting any remaining unfinished Work ("punch list" items). Substantial Completion of the Work, or a designated portion of the Work, is not achieved until so agreed in a Certificate of Substantial Completion signed by the Contractor and Owner.

(b) The Contractor shall notify the Owner in writing when it considers the Work, or a portion of the Work which the Owner has agreed to accept separately, to be

substantially complete and ready for a Final Inspection pursuant to Section 3.16. In this notification the Contractor shall identify any items remaining to be completed or corrected for Final Acceptance prior to final payment.

(c) Substantial Completion is achieved and a Final Inspection is appropriate only when a minimal number of punch list items exists and only a short period of time will be required to correct or complete them. Upon receipt of the Contractor's notice for a Final Inspection, the Owner will advise the Contractor in writing of any conditions of the Work which the Owner is aware do not constitute Substantial Completion; otherwise, a Final Inspection will proceed within a reasonable time after the Contractor's notice is given. However, the Owner will not be required to prepare lengthy listings of punch list items; therefore, if the Final Inspection discloses that Substantial Completion has not been achieved, the Owner may discontinue or suspend the inspection until the Contractor does achieve Substantial Completion.

- (d) Certificate of Substantial Completion.
 - When the Work or a designated portion of the Work is substantially complete, the Owner will prepare and sign a Certificate of Substantial Completion to be signed in order by the Contractor, and Owner.
 - (ii) When signed by all parties, the Certificate of Substantial Completion shall establish the Date of Substantial Completion which is the date upon which:
 - (1) The Work, or designated portion of the Work, is accepted by the Owner as being ready for occupancy,
 - (2) The Contractor's one-year and special warranties for the Work covered by the Certificate commence, unless stated otherwise in the Certificate (the one-year warranty for punch list items completed or corrected after the period allowed in the Certificate shall commence on the date of their Final Acceptance), and
 - (3) Owner becomes responsible for building security, maintenance, utility services, and insurance, unless stated otherwise in the Certificate.
 - (iii) The Certificate of Substantial Completion shall set the time within which the Contractor shall finish all items on the "punch list" accompanying the Certificate. The completion of punch list items shall be a condition precedent to Final Payment.

(e) The Date of Substantial Completion of the Work, as set in the Certificate of Substantial Completion of the Work or of the last completed portion of the Work, establishes the extent to which the Contractor is liable for Liquidated Damages, if any; however, should the Contractor fail to complete all punch list items within thirty days, or such other time as may be stated in the respective Certificate of Substantial Completion, the Contractor shall bear any expenses incurred by the Owner as a result of such failure to complete punch list items in a timely manner.

3.33 OCCUPANCY OR USE PRIOR TO COMPLETION.

(a) **Upon Substantial Completion**. Prior to completion of the entire Work, the Owner may occupy or begin utilizing any designated portion of the Work on the agreed Date of Substantial Completion of that portion of the Work.

(b) Before Substantial Completion.

- (i) The Owner shall not occupy or utilize any portion of the Work before Substantial Completion of that portion has been achieved.
- (ii) The Owner may deliver furniture and equipment and store, or install it in place ready for occupancy and use, in any designated portion of the Work before it is substantially completed under the following conditions:
 - (1) The Owner's storage or installation of furniture and equipment will not unreasonably disrupt or interfere with the Contractor's completion of the designated portion of the Work.
 - (2) The Contractor consents to the Owner's planned action (such consent shall not be unreasonably withheld).
 - (3) The Owner shall be responsible for insurance coverage of the Owner's furniture and equipment, and the Contractor's liability shall not be increased.
 - (4) The Contractor and Owner will jointly inspect and record the condition of the Work in the area before the Owner delivers and stores or installs furniture and equipment; the Owner will equitably compensate the Contractor for making any repairs to the Work that may subsequently be required due to the Owner's delivery and storage or installation of furniture and equipment.
 - (5) The Owner's delivery and storage or installation of furniture and equipment shall not be deemed an acceptance of any Work not completed in accordance with the requirements of the Contract Documents.

3.34 FINAL PAYMENT.

(a) **Prerequisites to Final Payment**. The following conditions are prerequisites to Final Payment becoming due the Contractor:

- (i) Full execution of a Certificate of Substantial Completion for the Work, or each designated portion of the Work.
- (ii) Final Acceptance of the Work.
- (iii) The Contractor's completion, to the satisfaction of Owner, of all documentary requirements of the Contract Documents; such as delivery of "as-built" documents, operating and maintenance manuals, warranties, etc.
- (iv) Delivery to the Owner of a final Application for Payment, prepared by the Contractor.
- (v) Completion of an Advertisement for Completion pursuant to Paragraph (c) below.

- (vi) Delivery by the Contractor to the Owner of a Release of Claims and such other documents as may be required by Owner, satisfactory in form to the Owner pursuant to Paragraph (d) below.
- (vii) Consent of Surety, if any, to Final Payment to Contractor.
- (viii) Delivery by the Contractor to the Owner of other documents, if any, required by the Contract Documents as prerequisites to Final Payment.

(b) Final Acceptance of the Work. "Final Acceptance of the Work" shall be achieved when all "punch list" items recorded with the Certificate(s) of Substantial Completion are accounted for by either:

- (i) Their completion or correction by the Contractor and acceptance by the Owner
- (ii) Their resolution under Section 3.18, Deductions for Uncorrected Work.

(c) Advertisement for Completion.

- (i) If the Contract Sum is less than \$50,000: The Owner, immediately after determining that all other requirements of the Contract have been completed, shall give public notice of completion of the Contract by having an Advertisement for Completion published one time in a newspaper of general circulation, published in the county in which the Owner is located and shall post notice of completion of the Contract on the Owner's bulletin board for one week, and shall require the Contractor to certify under oath that all bills have been paid in full. Final payment may be made at any time after the notice has been posted for one entire week.
- If the Contract Sum is more than \$50,000: The Contractor, (ii) immediately after being notified by the Owner that all other requirements of the Contract have been completed, shall give public notice of completion of the Contract by having an Advertisement for Completion published for a period of four successive weeks in some newspaper of general circulation published within the city or county where the Work was Proof of publication of the Advertisement for performed. Completion, in duplicate, shall be made by the Contractor to the Owner by affidavit of the publisher and a printed copy of the Advertisement for Completion published, in duplicate. If no newspaper is published in the county where the work was done, the notice may be given by posting at the Court House for thirty days and proof of same made by Probate Judge or Sheriff and the Contractor. Final payment shall not be due until thirty days after this public notice is completed.

(d) **Release of Claims**. The Release of Claims and other documents referenced in Paragraph (a) (vi) above are as follows:

(i) A release executed by Contractor of all claims and claims of lien against the Owner arising under and by virtue of the Contract,

other than such claims of the Contractor, if any, as may have been previously made in writing and as may be specifically excepted by the Contractor from the operation of the release in stated amounts to be set forth therein.

- (ii) An affidavit under oath, if required, stating that so far as the Contractor has knowledge or information, there are no claims or claims of lien which have been or will be filed by any Subcontractor, Supplier or other party for labor or material for which a claim or claim of lien could be filed.
- (iii) A release, if required, of all claims and claims of lien made by any Subcontractor, Supplier or other party against the Owner or unpaid Contract funds held by the Owner arising under or related to the Work on the Project; provided, however, that if any Subcontractor, Supplier or others refuse to furnish a release of such claims or claims of lien, the Contractor may furnish a bond executed by Contractor and its Surety to the Owner to provide an unconditional obligation to defend, indemnify and hold harmless the Owner against any loss, cost or expense, including attorney's fees, arising out of or as a result of such claims, or claims of lien, in which event Owner may make Final Payment notwithstanding such claims or claims of lien. If Contractor and Surety fail to fulfill their obligations to Owner under the bond, the Owner shall be entitled to recover damages as a result of such failure, including all costs and reasonable attorney's fees incurred to recover such damages.

(e) Effect of Final Payment.

- (i) The making of Final Payment shall constitute a waiver of Claims by the Owner except those arising from:
 - (1) Liens, claims, security interests or encumbrances arising out of the Contract and unsettled;
 - (2) Failure of the Work to comply with the requirements of the Contract Documents;
 - (3) Terms of warranties or indemnities required by the Contract Documents, or
 - (4) Blatant defects.
- (ii) Acceptance of Final Payment by the Contractor shall constitute a waiver of claims by Contractor except those previously made in writing, identified by Contractor as unsettled at the time of final Application for Payment, and specifically excepted from the release provided for in Paragraph (d) (1), above.

3.35 CONTRACTOR'S WARRANTY.

(a) General Warranty. The Contractor warrants to the Owner that all materials and equipment furnished under the Contract will be of good quality and new, except such materials as may be expressly provided or allowed in the Contract Documents to be otherwise, and that none of the Work will be Defective Work as defined in Section 3.1.

(b) **One-Year Warranty**.

- (i) If, within one year after the date of Completion of the Work or each designated portion of the Work (or otherwise as agreed upon in a mutually-executed Certificate of Completion), any of the Work is found to be Defective Work, the Contractor shall promptly upon receipt of written notice from the Owner and without expense to the Owner, replace or correct the Defective Work to conform to the requirements of the Contract Documents, and repair all damage to the site, the building and its contents which is the result of Defective Work or its replacement or correction.
- (ii) The one-year warranty for punch list items shall begin on the Date of Completion if they are completed or corrected within the time period allowed in the Certificate of Substantial Completion in which they are recorded. The one-year warranty for punch list items that are not completed or corrected within the time period allowed in the Certificate of Substantial Completion, and other Work performed after Substantial Completion, shall begin on the date of Final Acceptance of the Work. The Contractor's correction of Work pursuant to this warranty does not extend the period of the warranty. The Contractor's one-year warranty does not apply to defects or damages due to improper or insufficient maintenance, improper operation, or wear and tear during normal usage.
- (iii) Upon recognizing a condition of Defective Work, the Owner shall promptly notify the Contractor of the condition. If the condition is causing damage to the building, its contents, equipment, or site, the Owner shall take reasonable actions to mitigate the damage or its continuation, if practical. If the Contractor fails to proceed promptly to comply with the terms of the warranty, or to provide the Owner with satisfactory written verification that positive action is in process, the Owner may have the Defective Work replaced or corrected and the Contractor and the Contractor's Surety shall be liable for all expense incurred.
- (iii) Year-End Inspection(s): An inspection of the Work, or each separately completed portion thereof, is required near the end of the Contractor's one-year warranty period(s). The subsequent delivery of the report of a Year-end Inspection will serve as confirmation that the Contractor was notified of Defective Work found within the warranty period.
- (v) The Contractor's warranty of one year is in addition to, and not a limitation of, any other remedy stated herein or available to the Owner under applicable law.

3.36 INDEMNIFICATION AGREEMENT. To the fullest extent permitted by law, the Contractor shall defend, indemnify, and hold harmless the Owner, Owner's consultants, and their agents, employees, and consultants (hereinafter collectively referred to as the

"Indemnitees") from and against all claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of, related to, or resulting from performance of the Work, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property, including loss of use resulting there from, and is caused in whole or in part by negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, regardless of whether such claim, damage, loss or expense is caused in part, or is alleged but not legally established to have been caused in whole or in part by the negligence or other fault of a party indemnified hereunder.

(a) This indemnification shall extend to all claims, damages, losses and expenses for injury or damage to adjacent or neighboring property, or persons injured thereon, that arise out of, relate to, or result from performance of the Work.

(b) This indemnification does not apply to the extent of the sole negligence of the Indemnitees.

3.37 CONTRACTOR'S AND SUBCONTRACTORS' INSURANCE.

- (a) General.
 - (i) **Responsibility**. The Contractor shall be responsible to the Owner from the time of the signing of the Construction Contract or from the beginning of the first work, whichever shall be earlier, for all injury or damage of any kind resulting from any negligent act or omission or breach, failure or other default regarding the work by the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, regardless of who may be the owner of the property.
 - (ii) Insurance Providers. Each of the insurance coverage's required below shall be issued by an insurer licensed by the Insurance Commissioner to transact the business of insurance in the State of Alabama for the applicable line of insurance, and such insurer (or, for qualified self-insured or group self-insured, a specific excess insurer providing statutory limits) must have a Best Policyholders Rating of "A-" or better and a financial size rating of Class V or larger.
 - (iii) Notification Endorsement. Each policy shall be endorsed to provide that the insurance company agrees that the policy shall not be canceled, changed, allowed to lapse or allowed to expire for any reason until thirty days after the Owner has received written notice by certified mail as evidenced by return receipt or until such time as other insurance coverage providing protection equal to protection called for in the Contract Documents shall have been received, accepted and acknowledged by the Owner. Such notice shall be valid only as to the Project as shall have been designated by Project Name and Number in said notice.
 - (iv) **Insurance Certificates**. The Contractor shall procure the insurance coverage's identified below, or as otherwise required in the Contract Documents, at the Contractor's own expense, and to

evidence that such insurance coverage's are in effect, the Contractor shall furnish the Owner an insurance certificate(s) acceptable to the Owner and listing the Owner as the certificate holder. The insurance certificate(s) must be delivered to the Owner with the Construction Contract and Bonds for final approval and execution of the Construction Contract. The insurance certificate must provide the following:

- (1) Name and address of authorized agent of the insurance company
- (2) Name and address of insured
- (3) Name of insurance company or companies
- (4) Description of policies
- (5) Policy Number(s)
- (6) Policy Period(s)
- (7) Limits of liability
- (8) Name and address of Owner as certificate holder
- (9) Project Name and Number, if any
- (10) Signature of authorized agent of the insurance company
- (11) Telephone number of authorized agent of the insurance company
- (12) Mandatory thirty day notice of cancellation / non-renewal / change
- (v) **Maximum Deductible**. Self-insured retention, except for qualified self-insurers or group self-insurers, in any policy shall not exceed \$25,000.00.

(b) Insurance Coverage's. Unless otherwise provided in the Contract Documents, the Contractor shall purchase the types of insurance coverage's with liability limits not less than as follows:

(i) Workers' Compensation and Employer's Liability Insurance.

- (1) Workers' Compensation coverage shall be provided in accordance with the statutory coverage required in Alabama. A group insurer must submit a certificate of authority from the Alabama Department of Industrial Relations approving the group insurance plan. A selfinsurer must submit a certificate from the Alabama Department of Industrial Relations stating the Contractor qualifies to pay its own workers' compensation claims.
- (2) Employer's Liability Insurance limits shall be at least: Bodily Injury by Accident - \$1,000,000 each accident Bodily Injury by Disease - \$1,000,000 each employee

(ii) Commercial General Liability Insurance.

(1) Commercial General Liability Insurance, written on an ISO Occurrence Form (current edition as of the date of Advertisement for Bids) or equivalent, shall include, but need not be limited to, coverage for bodily injury and property damage arising from premises and operations liability, products and completed operations liability, blasting and explosion, collapse of structures, underground damage, personal injury liability and contractual liability. The Commercial General Liability Insurance shall provide at minimum the following limits:

	Coverage	Limit
1.	General Aggregate	\$1,000,000 per Project
2.	Products, Completed	\$1,000,000 per Project
	Operations Aggregate	
3.	Personal and Advertising	\$1,000,000 per Occurrence
	Injury	
4.	Each Occurrence	\$1,000,000

- (2) Additional Requirements for Commercial General Liability Insurance:
 - a. The policy shall name the Owner, and their agents, consultants and employees as additional insured, state that this coverage shall be primary insurance for the additional insured; and contain no exclusions of the additional insured relative to job accidents.
 - b. The policy must include separate per project aggregate limits.

(iii) Commercial Business Automobile Liability Insurance.

- (1) Commercial Business Automobile Liability Insurance which shall include coverage for bodily injury and property damage arising from the operation of any owned, nonowned or hired automobile. The Commercial Business Automobile Liability Insurance Policy shall provide not less than \$1,000,000 Combined Single Limits for each occurrence.
- (2) The policy shall name the Owner, and their agents, consultants, and employees as additional insureds.

(iv) Commercial Umbrella Liability Insurance.

- (1) Commercial Umbrella Liability Insurance to provide excess coverage above the Commercial General Liability, Commercial Business Automobile Liability and the Workers' Compensation and Employer's Liability to satisfy the minimum limits set forth herein.
- Minimum Combined Primary Commercial General Liability and Commercial/Excess Umbrella Limits of: \$ 2,000,000 per Occurrence \$ 2,000,000 Aggregate
- (3) Additional Requirements for Commercial Umbrella Liability Insurance:

- a. The policy shall name the Owner, and their agents, consultants, and employees as additional insureds.
- b. The policy must be on an "occurrence" basis.

(v) **Builder's Risk Insurance**.

- (1) The Builder's Risk Policy shall be made payable to the Owner and Contractor, as their interests may appear. The policy amount shall be equal to 100% of the Contract Sum, written on a Causes of Loss - Special Form (current edition as of the date of Advertisement for Bids), or its equivalent. All deductibles shall be the sole responsibility of the Contractor.
- (2) The policy shall be endorsed as follows: "The following may occur without diminishing, changing, altering or otherwise affecting the coverage and protection afforded the insured under this policy:
 - a. Furniture and equipment may be delivered to the insured premises and installed in place ready for use; or
 - b. Partial or complete occupancy by Owner; or
 - c. Performance of work in connection with construction operations insured by the Owner, by agents or lessees or other contractors of the Owner, or by contractors of the lessee of the Owner."

(c) Subcontractors' Insurance.

- (i) Workers' Compensation and Employer's Liability Insurance. The Contractor shall require each Subcontractor to obtain and maintain Workers' Compensation and Employer's Liability Insurance coverages as described in preceding Paragraph (b), or to be covered by the Contractor's Workers' Compensation and Employer's Liability Insurance while performing Work under the Contract.
- (ii) Liability Insurance. The Contractor shall require each Subcontractor to obtain and maintain adequate General Liability, Automobile Liability, and Umbrella Liability Insurance coverages similar to those described in preceding Paragraph (b). Such coverage shall be in effect at all times that a Subcontractor is performing Work under the Contract.
- (iii) **Enforcement Responsibility**. The Contractor shall have responsibility to enforce its Subcontractors' compliance with these or similar insurance requirements; however, the Contractor shall, upon request, provide the Owner acceptable evidence of insurance for any Subcontractor.

(d) **Termination of Obligation to Insure**. Unless otherwise expressly provided in the Contract Documents, the obligation to insure as provided herein shall continue as follows:

- (i) <u>Builder's Risk Insurance.</u> The obligation to insure under Subparagraph b(v) shall remain in effect until the Date of Substantial Completion as shall be established in the Certificate of Substantial Completion. In the event that multiple Certificates of Substantial Completion covering designated portions of the Work are issued, Builder's Risk coverage shall remain in effect until the Date of Substantial Completion as shall be established in the last issued Certificate of Substantial Completion. However, in the case that the Work involves separate buildings, Builder's Risk coverage of each separate building may terminate on the Date of Substantial Completion as established in the Certificate of Substantial Completion issued for each building.
- (ii) <u>Products and Completed Operations</u>. The obligation to carry Products and Completed Operations coverage specified under Subparagraph b(ii) shall remain in effect for two years after the Date(s) of Substantial Completion.
- (iii) <u>All other Insurance.</u> The obligation to carry other insurance coverage's specified under Paragraph (b) shall remain in effect after the Date(s) of Substantial Completion until such time as all Work required by the Contract Documents is completed. Equal or similar insurance coverages shall remain in effect if, after completion of the Work, the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, returns to the Project to perform warranty or maintenance work pursuant to the terms of the Contract Documents.

Waivers of Subrogation. The Owner and Contractor waive all rights (e) against (1) each other and any of their subcontractors, sub-subcontractors, agents and employees, each of the other, and (2) the separate contractors performing construction or operations related to the Project, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by builder's risk insurance or other property insurance applicable to the Work or to other property located within or adjacent to the Project, except such rights as they may have to proceeds of such insurance held by the Owner or Contractor as fiduciary. The Owner or Contractor, as appropriate, shall require of the separate contractors, if any, and the subcontractor, sub-subcontractors, suppliers, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The Policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to the person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged. The waivers provided for in this paragraph shall survive final acceptance and continue to apply to insured losses to the Work or other property on or adjacent to the Project.

3.38 PERFORMANCE AND PAYMENT BONDS.

(a) General. Upon signing and returning the Construction Contract to the Owner for final approval and execution, the Contractor shall, at the Contractor's expense, furnish to the Owner a Performance Bond and a Payment Bond, each in a penal sum equal to the amount of \$20,000. Each bond shall be on the form contained in the Bid Package, shall be executed by a surety company (Surety) acceptable to the Owner and duly authorized and qualified to make such bonds in the State of Alabama in the required amounts, shall be countersigned by an authorized, Alabama resident agent of the Surety who is qualified to execute such instruments, and shall have attached thereto a power of attorney of the signing official.

The provisions of this Section are not applicable to this Contract if the Contract Sum is less than \$50,000, unless bonds are required for this Contract in the Supplemental General Conditions.

(b) **Performance Bond**. Through the Performance Bond, the Surety's obligation to the Owner shall be to assure the prompt and faithful performance of the Contract and Contract Change Orders. The Penal Sum shall remain equal to the Contract Sum as the Contract Sum is adjusted by Contract Change Orders. In case of default on the part of the Contractor, the Surety shall take charge of and complete the Work in accordance with the terms of the Performance Bond. Any reasonable expenses incurred by the Owner as a result of default on the part of the Contractor, including architectural, engineering, administrative, and legal services, shall be recoverable under the Performance Bond.

(c) **Payment Bond**. Through the Payment Bond the Surety's obligation to the Owner shall be to guarantee that the Contractor and its Subcontractors shall promptly make payment to all persons supplying labor, materials, or supplies for, or in, the prosecution of the Work, including the payment of reasonable attorney's fees incurred by successful claimants or plaintiffs in civil actions on the Bond. Any person or entity indicating that they have a claim of nonpayment under the Bond shall, upon written request, be promptly furnished a certified copy of the Bond and Construction Contract by the Contractor, or Owner, whomever is recipient of the request.

(d) Change Orders. The Penal Sum shall remain equal to the Contract Sum as the Contract Sum is adjusted by Contract Change Orders. All Contract Change Orders involving an increase in the Contract Sum will require consent of Surety by endorsement of the Contract Change Order form. The Surety waives notification of any Contract Change Orders involving only extension of the Contract Time.

(e) **Expiration**. The obligations of the Contractor's performance bond surety shall be coextensive with the contractor's performance obligations under the Contract Documents; provided, however, that the surety's obligation shall expire at the end of the one-year warranty period(s) of Section 3.35.

3.39 ASSIGNMENT. The Contractor shall not assign the Contract or sublet it as a whole nor assign any moneys due or to become due to the Contractor thereunder without the previous written consent of the Owner (and of the Surety, in the case of a bonded Construction Contract). As prescribed by the Public Works Law, the Contract shall in no event be assigned to an unsuccessful bidder for the Contract whose bid was rejected because the bidder was not a responsible or responsive bidder.

3.40 CONSTRUCTION BY OWNER OR SEPARATE CONTRACTORS.

(a) **Owner's Reservation of Right**.

- (i) The Owner reserves the right to self-perform, or to award separate contracts for, other portions of the Project and other Project related construction and operations on the site. The contractual conditions of such separate contracts shall be substantially similar to those of this Contract, including insurance requirements and the provisions of this section. If the Contractor considers such actions to involve delay or additional cost under this Contract, notifications and assertion of claims shall be as provided in Section 3.20 and Section 3.23.
- (ii) When separate contracts are awarded, the term "Contractor" in the separate Contract Documents shall mean the Contractor who executes the respective Construction Contract.

(b) **Coordination**. Unless otherwise provided in the Contract Documents, the Owner shall be responsible for coordinating the activities of the Owner's forces and separate contractors with the Work of the Contractor. The Contractor shall cooperate with the Owner and separate contractors, shall participate in reviewing and comparing their construction schedules relative to that of the Contractor when directed to do so, and shall make and adhere to any revisions to the construction schedule resulting from a joint review and mutual agreement.

(c) Conditions Applicable to Work Performed by Owner. Unless otherwise provided in the Contract Documents, when the Owner self-performs construction or operations related to the Project, the Owner shall be subject to the same obligations to Contractor as Contractor would have to a separate contractor under the provision of this Section 3.40.

(d) Mutual Responsibility.

- (i) The Contractor shall reasonably accommodate the required introduction and storage of materials and equipment and performance of activities by the Owner and separate contractors and shall connect and coordinate the Contractor's Work with theirs as required by the Contract Documents.
- (ii) By proceeding with an element or portion of the Work that is applied to or performed on construction by the Owner or a separate contractor, or which relies upon their operations, the Contractor accepts the condition of such construction or operations as being suitable for the Contractor's Work, except for conditions that are not reasonably discoverable by the Contractor. If the Contractor discovers any condition in such construction or operations that is not suitable for the proper performance of the Work, the Contractor shall not proceed, but shall instead promptly notify the Owner in writing of the condition discovered.
- (iii) The Contractor shall reimburse the Owner for any costs incurred by a separate contractor and payable by the Owner because of acts or omissions of the Contractor. Likewise, the Owner shall be responsible to the Contractor for any costs incurred by the Contractor because of the acts or omissions of a separate contractor.

- (iv) The Contractor shall not cut or otherwise alter construction by the Owner or a separate contractor without the written consent of the Owner and separate contractor; such consent shall not be unreasonably withheld. Likewise, the Contractor shall not unreasonably withhold its consent allowing the Owner or a separate contractor to cut or otherwise alter the Work.
- (v) The Contractor shall promptly remedy any damage caused by the Contractor to the construction or property of the Owner or separate contractors.

3.41 SUBCONTRACTS.

(a) Award of Subcontracts and Other Contracts for Portion of the Work.

- Unless otherwise provided in the Contract Documents, when (i) delivering the executed Construction Contract, bonds, and evidence of insurance to the Owner the Contractor shall also submit a listing of Subcontractors proposed for each principal portion of the Work and fabricators or suppliers proposed for furnishing materials or equipment fabricated to the design of the Contract Documents. This listing shall be in addition to any naming of Subcontractors, fabricators, or suppliers that may have been required in the bid process. The Owner will promptly reply to the Contractor in writing stating whether or not the Owner, after due investigation, has reasonable objection to any Subcontractor, fabricator, or supplier proposed by the Contractor. The issuance of the Notice to Proceed in the absence of such objection by the Owner shall constitute notice that no reasonable objection to them is made. The Contractor shall not contract with a Subcontractor to perform more than 10% of the Work.
- (ii) The Contractor shall not contract with a proposed Subcontractor, fabricator, or supplier to whom the Owner has made reasonable and timely objection. Except in accordance with prequalification procedures as may be contained in the Contract Documents, through specified qualifications, or on the grounds of reasonable objection, the Owner may not restrict the Contractor's selection of Subcontractors, fabricators, or suppliers.
- (iii) Upon the Owner's reasonable objection to a proposed Subcontractor, fabricator, or supplier, the Contractor shall promptly propose another to whom the Owner has no reasonable objection. If the proposed Subcontractor, fabricator, or supplier to whom the Owner made reasonable objection was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be equitably adjusted by Contract Change Order for any resulting difference if the Contractor has acted promptly and responsively in this procedure.
- (iv) The Contractor shall not change previously selected Subcontractors, fabricators, or suppliers without notifying the

Owner in writing of proposed substitute Subcontractors, fabricators, or suppliers. If the Owner does not make a reasonable objection to a proposed substitute within three working days, the substitute shall be deemed approved.

(b) Subcontractor Relations.

- (i) The Contractor agrees to bind every Subcontractor and material supplier (and require every Subcontractor to so bind its subcontractors and material suppliers) to all the provisions of the Contract Documents as they apply to the Subcontractor's and material supplier's portion of the Work.
- (ii) Nothing contained in the Contract Documents shall be construed as creating any contractual relationship between any Subcontractor and the Owner, nor to create a duty of the Owner to resolve disputes between or among the Contractor or its Subcontractors and suppliers or any other duty to such Subcontractors or suppliers.

3.42 PERMITS, LAWS AND REGULATIONS.

(a) **Permits, Fees and Notices.**

- (i) Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit and other permits and governmental fees, licenses, and inspections necessary for proper execution and completion of the Work which are customarily secured after award of the Construction Contract and which are in effect on the date of receipt of bids.
- (ii) The Contractor shall comply with and give notices required by all laws, ordinances, rules, regulations, and lawful orders of public
- (b) **Compensation for Increases**. The Contractor shall be compensated for additional costs incurred because of increases in tax rates imposed after the date of receipt of bids.

3.43 ROYALTIES, PATENTS AND COPYRIGHTS. The Contractor shall pay all royalties and license fees. The Contractor shall defend, indemnify and hold harmless the Owner and their agents, employees, and consultants from and against all claims, damages, losses and expenses, including but not limited to attorney's fees, arising out of, related to, or resulting from all suits or claims for infringement of any patent rights or copyrights arising out of the inclusion of any patented or copyrighted materials, methods, or systems selected by the Contractor and used during the execution of or incorporated into the Work. This indemnification does not apply to any suits or claims of infringement of any patent rights or copyrights arising out of any patented or copyrighted materials, methods, or systems specified in the Contract Documents. However, if the Contractor has information that a specified material, method, or system is or may constitute an infringement of a patent or copyright, the Contractor shall be responsible for any resulting loss unless such information is promptly furnished to the Owner.

3.44 USE OF THE SITE.

(a) The Contractor shall confine its operations at the Project site to areas permitted by the Owner and by law, ordinances, permits and the Contract Documents and shall

not unreasonably encumber the site with materials, equipment, employees' vehicles, or debris. The Contractor's operations at the site shall be restricted to the sole purpose of constructing the Work, use of the site as a staging, assembly, or storage area for other business which the Contractor may undertake shall not be permitted.

(b) Unless otherwise provided in the Contract Documents, temporary facilities, such as storage sheds, shops, and offices may be erected on the Project site with the approval of the Owner. Such temporary buildings and/or utilities shall remain the property of the Contractor, and be removed at the Contractor's expense upon completion of the Work, unless the Owner authorizes their abandonment without removal.

3.45 CUTTING AND PATCHING.

(a) The Contractor shall be responsible for all cutting, fitting, or patching that may be required to execute the Work to the results indicated in the Contract Documents or to make its parts fit together properly.

(b) Any cutting, patching, or excavation by the Contractor shall be supervised and performed in a manner that will not endanger persons nor damage or endanger the Work or any fully or partially completed construction of the Owner or separate contractors.

3.46 IN-PROGRESS AND FINAL CLEANUP.

(a) In-Progress Clean-Up.

- (i) The Contractor shall at all times during the progress of the Work keep the premises and surrounding area free from rubbish, scrap materials and debris resulting from the Work. Trash and combustible materials shall not be allowed to accumulate inside buildings or elsewhere on the premises. At no time shall any rubbish be thrown from window openings. Burning of trash and debris on site is not permitted.
- (ii) The Contractor shall make provisions to minimize and confine dust and debris resulting from construction activities.

(b) Final Clean-Up. Before Substantial Completion or Final Acceptance is achieved, the Contractor shall have removed from the Owner's property all construction equipment, tools, and machinery; temporary structures and/or utilities including the foundations thereof (except such as the Owner permits in writing to remain); rubbish, debris, and waste materials; and all surplus materials, leaving the site clean and true to line and grade, and the Work in a safe and clean condition, ready for use and operation.

(c) **Owner's Right to Clean-Up**. If the Contractor fails to comply with these clean-up requirements and then fails to comply with a written directive by the Owner to clean-up the premises within a specified time, the Owner may implement appropriate clean-up measures and the cost thereof shall be deducted from any amounts due or to become due the Contractor.

3.47 LIQUIDATED DAMAGES.

(a) Time is the essence of the Contract. Any delay in the completion of the Work required by the Contract Documents may cause inconvenience to the public and loss and damage to the Owner including but not limited to interest and additional administrative, architectural, inspection and supervision charges. By executing the Construction Contract, the

Contractor agrees that the Contract Time is sufficient for the achievement of Substantial Completion.

(b) The Contract Documents may provide in the Construction Contract or elsewhere for a certain dollar amount for which the Contractor and its Surety (if any) will be liable to the Owner as liquidated damages for each calendar day after expiration of the Contract Time that the Contractor fails to achieve Substantial Completion of the Work. If such daily liquidated damages are provided for, Owner and Contractor, and it's Surety, agree that such amount is reasonable and agree to be bound thereby.

(c) If a daily liquidated damage amount is not otherwise provided for in the Contract Documents, a time charge equal to six percent interest per annum on the total Contract Sum may be made against the Contractor for the entire period after expiration of the Contract Time that the Contractor fails to achieve Substantial Completion of the Work.

(d) The amount of liquidated damages due under either paragraph b or c above, may be deducted by the Owner from the moneys otherwise due the Contractor in the Final Payment, not as a penalty, but as liquidated damages sustained, or the amount may be recovered from Contractor or its Surety. If part of the Work is substantially completed within the Contract Time and part is not, the stated charge for liquidated damages shall be equitably prorated to that portion of the Work that the Contractor fails to substantially complete within the Contract Time. It is mutually understood and agreed between the parties hereto that such amount is reasonable as liquidated damages.

3.48 USE OF FOREIGN MATERIALS.

(a) In the performance of the Work the Contractor agrees to use materials, supplies, and products manufactured, mined, processed or otherwise produced in the United States or its territories, if same are available at reasonable and competitive prices and are not contrary to any sole source specification implemented under the Public Works Law.

(b) In the performance of the Work the Contractor agrees to use steel produced in the United States if the Contract Documents require the use of steel and do not limit its supply to a sole source pursuant to the Public Works Law. If the Owner decides that the procurement of domestic steel products becomes impractical as a result of national emergency, national strike, or other cause, the Owner shall waive this restriction.

(c) If domestic steel or other domestic materials, supplies, and products are not used in accordance with preceding Paragraphs A and B, the Contract Sum shall be reduced by an amount equal to any savings or benefits realized by the Contractor.

(d) This Article applies only to Public Works projects financed entirely by the State of Alabama or any political subdivision of the state.

4. SPECIAL CONDITIONS

4.1 COORDINATION WITH WATER DEPARTMENT.

(a) Mr. Jay Craft, Madison County Water Department, will be the primary contact for the contract, telephone 256-532-1552, email jaycraft@madisoncountyal.gov.

(b) A work order will be created for each landscaping job. Work orders will be issued in groups of no less than five (5). The Contractor shall obtain the work orders from the Water Department Office at 107-B Wholesale Avenue, Huntsville, AL 35811 during business hours of 7:00 AM through 3:30 PM Monday through Friday, excluding holidays.

(c) The Contractor must work behind Madison County Water Department repair crew such that all issued work orders are 100% complete within one month of the service line or main break repair. The Contractor shall maintain a work order completion status of at least 75% on a biweekly basis. Completion will be measured from the date the work order was issued to the date the work order is complete. The Owner will evaluate Contractor performance on a biweekly and monthly schedule.

(d) The Owner is responsible for providing the Contractor the weekly list of landscaping repairs as shown in Madison County Water Department database. The list is intended to assist the Contractor in scheduling, status reporting and providing awareness of the amount of work ahead.

(e) A "call-back" is a high-priority work order that shall be repaired within 3 days from issuance. A call-back shall take precedence over existing work orders, and if more than one call-back has been issued, then call-backs shall be addressed in order from their issuance. The Owner will be responsible for informing the Contractor of a call-back. The Owner anticipates call-backs will comprise less than 10% of the total work orders issued.

(f) The work order completion requirement in Section 4.1(c) may be adjusted for an extension under extenuating weather conditions. Wet soil conditions that prohibit landscaping work or potentially compromise work performance may be considered an extenuating weather conditions and comply with Section 3.23, Delays. The Contractor is responsible for providing the Owner a notice of a potential delay in work. The delay, resulting in an adjustment to performance requirements in Section 4.1(c), will only be constituted if mutually agreed upon by the Owner and Contractor.

4.2 SOURCE OF PROJECT FUNDING. Local funds, deemed by the Owner to be an amount sufficient to fulfill the Owner's obligations under this Contract, are to be the source of funding for the project.

4.3 CONTRACT ADMINISTERED BY MADISON COUNTY WATER DEPARTMENT

(a) The contract will be administered by the Madison County Water Department. All correspondence, requests for information, application for payment should be sent to the Madison County Water Department in the attention of Mr. Jay Craft, Madison County Water Department, 107-B Wholesale Avenue, Huntsville, AL 35811.

(b) Contract change orders and supplemental agreements shall be signed by Mr. Dale W. Strong, Chairman of Madison County Commission.

4.4 OWNER'S REPRESENTATIVE. The Owner's Representative for this Project shall be: Mr. Jay Craft, Madison County Water Department, 107-B Wholesale Avenue, Huntsville, AL 35811, phone (256) 532-1552, email jaycraft@madisoncountyal.gov. The Owner's Representative shall review and approve progress payments in accordance with Section 3.29 (b). The Owner will notify the Contractor if the Owner's Representative or any associated contact information changes during the Project.

4.5 SPECIAL PROJECTS. The majority work performed under this contract will be landscaping repairs from service line and water main repairs. During the course of the contract, the Contractor may be required to perform landscaping not associated with a service line or main break repair. This work will be handled as a special project. The Madison County Water Department Representative and the Contractor will mutually agree on the scope of each special project. The special project work will be paid at an hourly or quantity unit price rate.

4.6 PAYMENT AND INVOICES. The Contractor shall be paid based on the unit prices contained in the construction contract. The invoice will be reviewed and approved by the owner's representative.

(a) The Contractor will be paid once a month for all work completed and approved by the Madison County Water Department during the month. The Contractor's invoice must include an entry for each address where landscaping was performed. For each entry, the invoice must identify the type of landscaping repair (short side service, long side service or main), the type of vegetation (seed or sod) and any additional installation of sod, seed or topsoil. For entries requiring Excess Yards, the dimensions of the repairs must be provided.

(b) The invoice must show the maximum contract sum, sum of previous invoices, amount of current invoice, and amount remaining on contract.

- (c) A schedule of values as specified in Paragraph 3.29 (c) is not required.
- (d) No retainage will be held on this contract.
- (e) The following Contractor certification will be included on all invoices.

"The Contractor certifies that, to the best of the Contractor's knowledge, information, and belief, the Work covered by the invoice has been completed in accordance with the Contract Documents, that all amounts have been paid by the Contractor for Work for which previous invoices were issued, and payments received from the Owner, and that the current payment shown in the invoice has not yet been received."

4.7 MATERIALS SUPPLIED BY THE CONTRACTOR. The Contractor shall furnish all labor, equipment, and landscaping materials, including topsoil, seed, straw, and sod. All seed must be of a blend that will grow during the season that it is applied. Sod shall be of the same type as the sod surrounding the disturbed area. All recommended seed and sod will require approval of a Madison County Water Department Representative (MCR).

4.8 CONTRACTOR RESPONSIBLE FOR LOCATING UTILITIES. The Contractor will be responsible for locating all underground utilities including but not limited to water mains, electric, telephone, natural gas, cable television, and fiber optic cables. The Contractor is responsible to insure all underground utilities are located by the proper

organizations and clearly marked on the ground. Additionally, the Contractor must insure the utility locates are kept current and updated during the Work.

4.9 DAMAGE TO PROPERTY. The Contractor shall pay at his own expense for damage to property resulting from the Work. Refer to Section 3.14 of the "General Conditions of the Contract"

4.10 TRAFFIC CONTROL. The Contractor is responsible for controlling traffic at the work sites. The Contractor shall provide and employ traffic control devices to minimize the inconvenience to motorist while maximizing safety according to the Manual on Uniform Traffic Control Devices, for Streets and Highways, Temporary Traffic Control, latest edition. Refer to Section 3.14, Safety and Protection of Persons and Property. The Contractor at no time during the construction shall close roads to vehicle traffic.

4.11 NO STAGING AREA AVAILABLE. The Contractor must make arrangements for the staging of materials and equipment.

4.12 BUILDER'S RISK INSURANCE NOT REQUIRED. Builder's Risk Insurance as described in Section 3.37 (b) (v) is not required for this project

5. TECHNICAL SPECIFICATIONS

5.1 GENERAL. The Contractor shall perform the following work per landscaping repair: clear the disturbed area of all debris (rocks, roots, dirt clods, etc.), return disturbed area to original grade using topsoil that has been screened and free of debris, compact the soil, seed and straw the disturbed area with seed that will properly grow during the season the repair is being made, and/or sod the disturbed area with the same sod existing around the disturbed area. The Contractor shall obtain approval from the Owner prior to installing any additional materials outside the scope of a typical landscaping repair.

5.2 SPECIFICATIONS. All assigned disturbed areas shall be seeded and strawed within 30 days of assignment. Disturbed areas to be seeded or sodded shall be brought to the designated grade with no less than the original depth of select topsoil, with due consideration being given to shrinkage and settlement. The Contractor shall be responsible for any additional landscaping required as a result of settlement. Topsoil shall be free of rocks, roots and debris. Imported topsoil shall be fertile sandy loam that will support vegetative growth. The Contractor shall compact the soil using the appropriate compaction equipment necessary for the job, such as heavy machinery or a hand tamper. A hand rake shall be used to smooth the topsoil prior to seeding and strawing.

The Contractor shall ensure that all repair sites will properly drain upon completion of landscaping. No grading shall result in the ditch below the invert of the culvert. All curb and gutter areas in subdivisions will be sodded. Contractor shall obtain Owner approval prior to installing any rip rap. Geotextile fabric is required to underlay rip rap to prevent scouring and improve the effectiveness of rip rap; the fabric shall be staked. All areas with steep slopes shall have protective netting installed or shall be sodded to establish vegetation and prevent erosion.

5.3 SITE CLEANUP. The Contractor shall at all times during the progress of the work keep the premises and surrounding area free from rubbish, scrap materials and debris resulting from the work. The site shall remain a safe and clean working environment. In no case shall any excess materials from the landscaping repair be left on site. The Contractor shall remove and dispose of any flags or caution tape in the repair area. Contractor shall ensure the roadway is free of mud and excess dirt after the repair is complete.

A.1 BID FORM

To:	Madison County Commission,	Date:
-----	----------------------------	-------

In compliance with your Advertisement for Bids and subject to all the conditions thereof, the undersigned (Legal Name of Bidder)

hereby proposes to furnish all labor and materials and perform all work required for the Work

Landscaping for Madison County Water Department, Bid 2017-54, dated September 2017

The Bidder, which is organized and existing under the laws of the State of ______, having its principal offices in the City of ______, is:

___Corporation ____Partnership ____Individual ____Other

LISTING OF PARTNERS OR OFFICERS: If Bidder is a Partnership, list all partners and their addresses; if Bidder is a Corporation, list the names, titles, and business addresses of its officers:

BIDDER'S REPRESENTATION: The Bidder declares that it has examined the site of the Work, having become fully informed regarding all pertinent conditions, and that it has examined the Specifications (including all Addenda received) for the Work and the other Bid and Contract Documents relative thereto, and that it has satisfied itself relative to the Work to be performed.

ADDENDA: The Bidder acknowledges receipt of Addenda Numbers._____through ______through ______

Bid Calculation Table

Instructions for Completion: Multiply Unit Price by Quantity to calculate Total.

Item Number	Item	Description	Unit	Unit Price	Quantity	Total
Landscapi	ng Repairs after Ser	vice Line Repairs and Main Break Repairs				
1	Landscaping	Landscaping sodded area after water main repair	EA		25	
2	Landscaping	Landscaping area with seed and straw after water main repair	EA		25	
3	Landscaping	Landscaping sodded area after long side service repair	EA		25	
4	Landscaping	Landscaping area with seed and straw after long side service repair	EA		25	
5	Landscaping	Landscaping sodded area after short side service repair	EA		25	
6	Landscaping	Landscaping area with seed and straw after short side service repair	EA		25	
Landscapi	ng Repairs and Spe	cial Projects				
7	Bermuda Sod	For providing and installing Bermuda sod	SY		1	
8	Seed and Straw (Bermuda)	For providing and installing Bermuda seed with straw covering	SY		1	
9	Topsoil	For providing and installing topsoil (4" thick)	CY		1	
Special Pro	ojects				1 1	
10	Skid Steer Loader	For the use of one (1) skid steer loader and operator per hour use at work location	HR		1	
11	Mini-excavator	For the use of one (1) mini-excavator and operator per hour use at work location	HR		1	
12	Dump Truck	For the use of one (1) dump truck and operator per hour use at work location	HR		1	
13	Laborer	For the use of one (1) laborer per hour use at work location	HR		1	
14	Rip Rap	For installing rip rap provided by Madison County	TON		1	
15	Removal of Miscellaneous Material	For removal and disposal of unclassified borrow and waste material	СҮ		1	
		ΤΟΤΑΙ	L =			

BID AMOUNT:

For the Work complete as specified using the bid quantities, the sum of ______

_____ Dollars (\$______) is bid.

BID SECURITY: The undersigned agrees to enter into a Construction Contract and furnish the prescribed Performance and Payment Bonds and evidence of insurance within fifteen calendar days, or such other period stated in the Bid Documents, after the contract forms have been presented for signature, provided such presentation is made within 60 calendar days after the opening of bids, or such other period stated in the Bid Documents. As security for this condition, the undersigned further agrees that the funds represented by the Bid Bond (or cashier's check) attached hereto may be called and paid into the account of the Awarding Authority as liquidated damages for failure to so comply.

Attached hereto is a: (Mark the appropriate box and provide the applicable information.)

Bid Bond, executed by		as Surety,
cashier's check on the	Bank	,
for the sum of		Dollars

(\$_____) made payable to the Awarding Authority.

BIDDER'S ALABAMA GENERAL CONTRACTORS LICENSE:

License Number

Bid Limit

Type(s) of Work

CERTIFICATIONS: The undersigned certifies that he or she is authorized to execute contracts on behalf of the Bidder as legally named, that this proposal is submitted in good faith without fraud or collusion with any other bidder, that the information indicated in this document is true and complete, and that the bid is made in full accord with State law. Notice of acceptance may be sent to the undersigned at the address set forth below.

The Bidder also declares that a list of all proposed major subcontractors and suppliers will be submitted at a time subsequent to the receipt of bids as established by the Owner in the Bid Documents but in no event shall this time exceed twenty-four (24) hours after receipt of bids.

I hereby certify that I have read and understand Madison County's general terms and conditions and affirm that I have not been in any agreement or collusion among bidders or prospective bidders in restraint of freedom of competition. Upon award of this bid, I will not substitute any item on this bid under any circumstances. I also understand that the general terms and conditions are standard and that any contradicting requirements of the IFB or RFP supersede.

Legal Name of Bidder	
Mailing Address	
* By (Legal Signature)	
* Name (type or print)	
*Title	

* Telephone Number

* If other than the individual proprietor, or an above named member of the Partnership, or the above named president, vice-president, or secretary of the Corporation, attach written authority to bind the Bidder. Any modification to a bid shall be over the initials of the person signing the bid, or of an authorized representative.

A.2 BID BOND

The PRINCIPAL:

The SURETY:

The OWNER: Madison County, Alabama, Madison County Courthouse, 100 Northside Square, Huntsville, Alabama 35801

The PROJECT for which the Principal's Bid is submitted:

Landscaping for Madison County Water Department, Bid 2017-54, dated September 2017

KNOW ALL MEN BY THESE PRESENTS, that we, the undersigned Principal and Surety, jointly and severally, hereby bind ourselves, our heirs, executors, administrators, successors, and assigns to the Owner in the PENAL SUM of five percent (5%) of the amount of the Principal's bid.

THE CONDITION OF THIS OBLIGATION is that the Principal has submitted to the Owner the attached bid, which is incorporated herein by reference, for the Project identified above.

NOW, THEREFORE, if, within the terms of the Bid Documents, the Owner accepts the Principal's bid and the Principal thereafter either:

(a) executes and delivers a Construction Contract with the required Performance and Payment Bonds (each in the form contained in the Bid Documents and properly completed in accordance with the bid) and delivers evidence of insurance as prescribed in the Bid Documents, or

(b) fails to execute and deliver such Construction Contract with such Bonds and evidence of insurance, but pays the Owner the difference, not to exceed the Penal Sum of this Bond, between the amount of the Principal's Bid and the larger amount for which the Owner may award a Construction Contract for the same Work to another bidder, then, this obligation shall be null and void, otherwise it shall remain in full force and effect.

The Surety, for value received, hereby stipulates and agrees that the obligation of the Surety under this Bond shall not in any manner be impaired or affected by any extension of the time within which the Owner may accept the Principal's bid, and the Surety does hereby waive notice of any such extension.

SIGNED AND SEALED this	day of	
ATTEST:	PRINCIPAL:	
	By	
	Name and Title	
	SURETY:	
ATTEST		
	By	
	Name and Title	

_.

A.3 UNAUTHORIZED ALIENS AFFIDAVIT

EACH VENDOR SUBMITTING A BID MUST COMPLETE AND SUBMIT WITH ITS BID THE FOLLOWING STATEMENT

STATE OF ALABAMA MADISON COUNTY

VERIFIED STATEMENT REGARDING UNAUTHORIZED ALIENS

The undersigned hereby certifies, under oath, in regard to the performance of the services or for the supply of materials or things described in this Invitation for Bid, that:

(A) it has conducted a verification, pursuant to all federal and state laws, of all the employees who will perform work on the Madison County Commission contract or work on the materials or things supplied to the Madison County Commission in response to this Invitation for Bid to insure that no unauthorized aliens will be employed to perform Madison County Commission work or supply materials or things to the Madison County Commission,

and

(B) to the best of their knowledge and belief, it is not employing or otherwise using unauthorized aliens to provide services, materials or things to the Madison County Commission.

The undersigned agrees:

1. it will verify that whether an employee is an unauthorized alien by inspecting such documents as are designated by Federal Law. For contracts in excess of \$100,000.00, in any twelve (12) month period of time, the contractor or supplier shall certify to the Commission that it has and will verify, to the extent allowable by Federal Law, by using the Federal E-Verify program, that no unauthorized aliens are utilized in providing services, materials or things to the Commission;

2. upon request, it will certify to the Commission, under oath by an officer or a management level employee, that it has verified to the extent allowable under Federal Law that named or otherwise described employees utilized in providing services, materials or things to the Commission are not unauthorized aliens;

3. upon determination by any appropriate Federal Agency that an employee is an unauthorized alien, the undersigned shall terminate the unauthorized alien's employment.

4. if the undersigned fails to terminate an employee determined to be an unauthorized alien by the Federal government or fails to provide the verification described above, the Madison County Commission may terminate the contract for the performance of services, materials or things pursuant to Madison County, Alabama, Policy Regarding The Employment of Unauthorized Aliens by Contractors and Suppliers.

The requirements and obligations of this Policy and Statement shall be interpreted and implemented in a manner consistent with all Federal and State Laws. If any provision of this Policy or Statement is declared invalid or in conflict with Federal or State Laws, such invalidly or conflict shall not affect the other provisions of this Policy or Statement which can be given effect without the invalid provision. The provisions of this Policy and Statement are declared to be severable.

NAI	ЛЕ:
BY:	
Prin	ted Name of Person Signing:
Pos	tion:
SWORN TO and subscrib	ed before me on this theday of, 20
	Notary Public
	My Commission Expires:

A.4 SWORN AFFIDAVIT OF EMPLOYER REGARDING UNATHORIZED ALIENS

1. The undersigned hereby attests by this sworn affidavit signed before a notary that it does not knowingly employ, hire for employment, or continue to employ an unauthorized alien.

2. The undersigned hereby attests that it is enrolled in the Federal E-Verify program. A true and correct copy of the undersigned's fully executed Federal E-Verify Memorandum of Understanding is attached hereto.

3. The Federal E-Verify Memorandum of Understanding attached hereto is the operative Memorandum of Understanding under which the undersigned is currently participating in the Federal E-Verify Program. **Must provide E-Verify Company ID:**

4. This sworn affidavit is provided to comply with Alabama Act No. 2011-535, known as the "Beason-Hammon Alabama Taxpayer and Citizen Protection Act."

Name:			
By:			
Printed Name:			

Position:

STATE OF ALABAMA

COUNTY OF MADISON

I, the undersigned, Notary Public in and for said County in said State, hereby certify that

______, whose name as _______ of ______, is signed to the foregoing instrument and who is known to me, acknowledged before me on this day that, being informed of the contents of the instrument, he, as such _______ and with full authority, executed the same voluntarily for and as the act of said ______.

Given under my hand this the _____day of _____, 20___.

Notary Public

My Commission Expires: _____

A.5. SWORN AFFIDAVIT OF SUBCONTRACTOR REGARDING UNATHORZIED ALIENS

1. The undersigned subcontractor hereby attests by this sworn affidavit signed before a notary that it does not knowingly employ, hire for employment, or continue to employ an unauthorized alien.

2. The undersigned subcontractor hereby attests that it is enrolled in the Federal E-Verify program. A true and correct copy of the undersigned's fully executed Federal E-Verify Memorandum of Understanding is attached hereto.

3. The Federal E-Verify Memorandum of Understanding attached hereto is the operative Memorandum of Understanding under which the undersigned is currently participating in the Federal E- Verify Program. **Must provide E-Verify Company ID:**

4. This sworn affidavit is provided to comply with Alabama Act No. 2011-535, known as the "Beason-Hammon Alabama Taxpayer and Citizen Protection Act."

Name:	
By:	
Printed Name:	
Position:	
STATE OF ALABAI	ЛА
COUNTY OF MADISC	N

I, the undersigned, Notary Public in and for said County in said State, hereby certify that

	, whose name as	of		,	is
signed	o the foregoing instrument and who is known	own to me, acknowledged b	efore me	on this	s day
that, be	ng informed of the contents of the instrum	ent, he, as such	and	with	full
authori	y, executed the same voluntarily for and as	s the act of said	·		

Given under my hand this the ______day of ______, 20____.

Notary Public

My Commission Expires:

B.1 CONSTRUCTION CONTRACT

This Construction Contact is entered into this 6th day of September 2017, between Madison County, Alabama ("Owner"), Madison County Courthouse, 100 Northside Square, Huntsville, Alabama 35801 and ("Contractor") for the WORK of the Project, identified as:

Landscaping for Madison County Water Department, Bid 2017-XX

The CONTRACT DOCUMENTS are dated September 2017 and have been amended by:

ADDENDA: NONE

The CONTRACT SUM:

The CONTRACT DURATION:

UNIT PRICES:

Item Number	Item	Description	Unit	Unit Price	Quantity	Total
Landscapir	ng Repairs after Ser	vice Line Repairs and Main Break Repairs				
1	Landscaping	Landscaping sodded area after water main repair	EA		25	
2	Landscaping	Landscaping area with seed and straw after water main repair	EA		25	
3	Landscaping	Landscaping sodded area after long side service repair	EA		25	
4	Landscaping	Landscaping area with seed and straw after long side service repair	EA		25	
5	Landscaping	Landscaping sodded area after short side service repair	EA		25	
6	Landscaping	Landscaping area with seed and straw after short side service repair	EA		25	
Landscapir	ng Repairs and Spe	cial Projects				
7	Bermuda Sod	For providing and installing Bermuda sod	SY		1	
8	Seed and Straw (Bermuda)	For providing and installing Bermuda seed with straw covering	SY		1	
9	Topsoil	For providing and installing topsoil (4" thick)	CY		1	
Special Pro	ojects		•	•		
10	Skid Steer Loader	For the use of one (1) skid steer loader and operator per hour use at work location	HR		1	
11	Mini-excavator	For the use of one (1) mini-excavator and operator per hour use at work location	HR		1	
12	Dump Truck	For the use of one (1) dump truck and operator per hour use at work location	HR		1	
13	Laborer	For the use of one (1) laborer per hour use at work location	HR		1	
14	Rip Rap	For installing rip rap provided by Madison County	TON		1	

15	Removal of	For removal and disposal of unclassified	CY	1	
	Miscellaneous	borrow and waste material			
	Material				

THE OWNER AND THE CONTRACTOR AGREE AS FOLLOWS:

The Contract Documents dated September 2017 are incorporated herein by reference. The Contractor shall perform the Work in accordance with the Contract Documents. The Owner will pay and the Contractor will accept as full compensation for such performance of the Work, the Contract Sum subject to actual quantities of unit priced work performed. The Work shall be commenced on a date to be specified in the Notice to Proceed.

The Owner and Contractor have entered into this Construction Contract as of the date first written above and have executed this Construction Contract in sufficient counterparts to enable each contracting party to have an originally executed Construction Contract each of which shall, without proof or accounting for the other counterparts, be deemed an original thereof.

The Owner does hereby certify that this Construction Contract was let in accordance with the provisions of Title 39, Code of Alabama 1975, as amended, and all other applicable provisions of law.

MADISON COUNTY, ALABAMA

Dale W. Strong, Chairman Madison County Commission

CONTRACTOR

By:

Its:

B.2 PAYMENT BOND:

The PRINCIPAL

The SURETY

The OWNER

Madison County, Alabama, Madison County Courthouse, 100 Northside Square, Huntsville, Alabama 35801

The PENAL SUM of this Bond:

Twenty Thousand Dollars (\$20,000)

DATE of the Construction Contract:

The PROJECT:

Landscaping for the Madison County Water Department, dated September 2017

1. WE, THE PRINCIPAL (hereinafter "Contractor") AND THE SURETY, jointly and severally, hereby bind ourselves, our heirs, executors, administrators, successors, and assigns to the Owner in the Penal Sum stated above to promptly pay all persons supplying labor, materials, or supplies for or in the prosecution of the Contract, which is incorporated herein by reference, and any modifications thereof by Contract Change Orders. If the Contractor and its Subcontractors promptly pay all persons supplying labor, materials, or supplies for or in the prosecution of the Contract Change Orders. If the prosecution of the Contract and Contract Change Orders, then this obligation shall be null and void; otherwise to remain and be in full force and effect.

2. The Penal Sum shall remain equal to\$20,000. All Contract Change Orders involving an increase in the Contract Sum will require consent of Surety by endorsement of the Contract Change Order form. The Surety waives notification of any Contract Change Orders involving only extension of the Contract Time.

3. Any person that has furnished labor, materials, or supplies for or in the prosecution of the Contract and Contract Change Orders for which payment has not been timely made may institute a civil action upon this Bond and have their rights and claims adjudicated in a civil action and judgment entered thereon. Notwithstanding the foregoing, a civil action may not be instituted on this bond until 45 days after written notice to the Surety of the amount claimed to be due and the nature of the claim. The civil action must commence not later than one year from the date of final settlement of the Contract. The giving of notice by registered or certified mail, postage prepaid, addressed to the Surety at any of its places of business or offices shall be deemed sufficient. In the event the Surety or Contractor fails to pay the claim in full within 45 days from the mailing of the notice, then the person or persons may recover from the Contractor and Surety, in addition to the amount of the claim, a reasonable attorney's fee based on the result, together with interest on the claim from the date of the notice.

4. Every person having a right of action on this bond shall, upon written application to the Owner indicating that labor, material, or supplies for the Work have been supplied and that payment has not been made, be promptly furnished a certified copy of this bond and the Construction Contract. The claimant may bring a civil action in the claimant's name on this Bond against the Contractor and the Surety, or

either of them, in the county in which the Work is to be or has been performed or in any other county where venue is otherwise allowed by law.

5. This bond is furnished to comply with Code of Alabama, §39-1-1, and all provisions thereof shall be applicable to civil actions upon this bond.

6. All claims and disputes between Owner and either the Contractor or Surety arising out of or related to this bond, or its breach, shall be resolved in accordance with Section 3.24, General Conditions of the Contract

SIGNED AND SEALED this	day of	,
ATTEST: CONTRACTOR as	PRINCIPAL:	
		ByName and Title
Countersigned by		
Alabama Resident Agent for Surety:		Surety:
By		
Name		Ву
Address		Name and Title

NOTE: Power of attorney for the Surety's signatory shall be furnished with the original and three copies of the bond.

B.3 PERFORMANCE BOND:

The PRINCIPAL

The SURETY

The OWNER

Madison County, Alabama, Madison County Courthouse, 100 Northside Square, Huntsville, Alabama 35801

The PENAL SUM of this Bond:

Twenty Thousand Dollars (\$20,000)

DATE of the Construction Contract:

The PROJECT:

Landscaping for the Madison County Water Department, dated September 2017

1. WE, THE PRINCIPAL (hereinafter "Contractor") AND THE SURETY, jointly and severally, hereby bind ourselves, our heirs, executors, administrators, successors, and assigns to the Owner in the Penal Sum stated above for the performance of the Contract, and Contract Change Orders, in accord with the requirements of the Contract Documents, which are incorporated herein by reference. If the Contractor performs the Contract, and Contract Change Orders, in accordance with the Contract Documents, then this obligation shall be null and void; otherwise it shall remain in full force and effect.

2. The Penal Sum shall remain equal to \$20,000. All Contract Change Orders involving an increase in the Contract Sum will require consent of Surety by endorsement of the Contract Change Order form. The Surety waives notification of any Contract Change Orders involving only extension of the Contract Time.

3. Whenever the Owner gives the Contractor and the Surety, at their addresses stated above, a written Notice to Cure a condition for which the Contract may be terminated in accordance with the Contract Documents, the Surety may, within the time stated in the notice, cure or provide the Owner with written verification that satisfactory positive action is in process to cure the condition.

4. The Surety's obligation under this Bond becomes effective after the Contractor fails to satisfy a Notice to Cure and the Owner:

(a) gives the Contractor and the Surety, at their addresses stated above, a written Notice of Termination declaring the Contractor to be in default under the Contract and stating that the Contractor's right to complete the Work, or a designated portion of the Work, shall terminate seven days after the Contractor's receipt of the notice; and

(b) gives the Surety a written demand that, upon the effective date of the Notice of Termination, the Surety promptly fulfill its obligation under this Bond.

5. In the presence of the conditions described in Paragraph 4, the Surety shall, at its expense:

(a) On the effective date of the Notice of Termination, take charge of the Work and be responsible for the safety, security, and protection of the Work, including materials and equipment stored on and off the Project site, and

(b) Within twenty-one days after the effective date of the Notice of Termination, proceed, or provide the Owner with written verification that satisfactory positive action is in process to

facilitate proceeding promptly, to complete the Work in accordance with the Contract Documents, either with the Surety's resources or through a contract between the Surety and a qualified contractor to whom the Owner has no reasonable objection.

6. As conditions precedent to taking charge of and completing the Work pursuant to Paragraph 5, the Surety shall neither require, nor be entitled to, any agreements or conditions other than those of this Bond and the Contract Documents. In taking charge of and completing the Work, the Surety shall assume all rights and obligations of the Contractor under the Contract Documents; however, the Surety shall also have the right to assert "Surety Claims" to the Owner in accordance with the Contract Documents. The presence or possibility of a Surety Claim shall not be just cause for the Surety to fail or refuse to promptly take charge of and complete the Work or for the Owner to fail or refuse to continue to make payments in accordance with the Contract Documents.

7. By accepting this Bond as a condition of executing the Construction Contract, and by taking the actions described in Paragraph 4, the Owner agrees that:

(a) the Owner shall promptly advise the Surety of the unpaid balance of the Contract Sum and, upon request, shall make available or furnish to the Surety, at the cost of reproduction, any portions of the Project Record, and

(b) as the Surety completes the Work, or has it completed by a qualified contractor, the Owner shall pay the Surety, in accordance with terms of payment of the Contract Documents, the unpaid balance of the Contract Sum, less any amounts that may be or become due the Owner from the Contractor under the Construction Contract or from the Contractor or the Surety under this Bond.

8. In the presence of the conditions described in Paragraph 4, the Surety's obligation includes responsibility for the correction of Defective Work, liquidated damages, and reimbursement of any reasonable expenses incurred by the Owner as a result of the Contractor's default under the Contract, including architectural, engineering, administrative, and legal services.

9. Nothing contained in this Bond shall be construed to mean that the Surety shall be liable to the Owner for an amount exceeding the Penal Sum of this Bond, except in the event that the Surety should be in default under the Bond by failing or refusing to take charge of and complete the Work pursuant to Paragraph 5. If the Surety should fail or refuse to take charge of and complete the Work, the Owner shall have the authority to take charge of and complete the Work, or have it completed, and the following costs to the Owner, less the unpaid balance of the Contract Sum, shall be recoverable under this Bond:

(a) the cost of completing the Contractor's responsibilities under the Contract, including correction of Defective Work;

(b) additional architectural, engineering, managerial, and administrative services, and reasonable attorneys' fees incident to completing the Work;

(c) interest on, and the cost of obtaining, funds to supplement the unpaid balance of the Contract Sum as may be necessary to cover the foregoing costs;

(d) the fair market value of any reductions in the scope of the Work necessitated by insufficiency of the unpaid balance of the Contract Sum and available supplemental funds to cover the foregoing costs; and

(f) additional architectural, engineering, managerial, and administrative services, and reasonable attorneys' fees incident to ascertaining and collecting the Owner's losses under the Bond.

10. All claims and disputes arising out of or related to this bond, or its breach, shall be resolved in accordance with Section 3.24, General Conditions of the Contract.

SIGNED AND SEALED this	_ day of	,
ATTEST:		Contractor as Principal:
		By
		Name and Title
Countersigned by Alabama Resident Agent for Surety:		SURETY:
By		
Name		By
Address	-	Name and Title

NOTE: Power of attorney for the Surety's signatory shall be furnished with the original and three copies of the bond